## MONTE CARLO WITH DETERMINANTAL POINT PROCESSES

BY RÉMI BARDENET<sup>1</sup> AND ADRIEN HARDY<sup>2</sup>

<sup>1</sup>Université de Lille, CNRS, Centrale Lille, UMR 9189 – CRIStAL, remi.bardenet@gmail.com <sup>2</sup>Université de Lille, CNRS, Inria, UMR 8524 – Laboratoire Paul Painlevé, adrien.hardy@univ-lille.fr

We show that repulsive random variables can yield Monte Carlo methods with faster convergence rates than the typical  $N^{-1/2}$ , where N is the number of integrand evaluations. More precisely, we propose stochastic numerical quadratures involving determinantal point processes associated with multivariate orthogonal polynomials, and we obtain root mean square errors that decrease as  $N^{-(1+1/d)/2}$ , where d is the dimension of the ambient space. First, we prove a central limit theorem (CLT) for the linear statistics of a class of determinantal point processes, when the reference measure is a product measure supported on a hypercube, which satisfies the Nevai-class regularity condition; a result which may be of independent interest. Next, we introduce a Monte Carlo method based on these determinantal point processes, and prove a CLT with explicit limiting variance for the quadrature error, when the reference measure satisfies a stronger regularity condition. As a corollary, by taking a specific reference measure and using a construction similar to importance sampling, we obtain a general Monte Carlo method, which applies to any measure with continuously derivable density. Loosely speaking, our method can be interpreted as a stochastic counterpart to Gaussian quadrature, which at the price of some convergence rate, is easily generalizable to any dimension and has a more explicit error term.

#### CONTENTS

1.	Introduction	369					
	1.1. Gaussian quadrature	369					
	1.2. Monte Carlo methods	370					
	1.3. Quasi-Monte Carlo methods	371					
	1.4. Bayesian quadrature	372					
	1.5. Our contribution	372					
2.	Statement of the results	374					
	2.1. Determinantal point processes and multivariate OP Ensembles	374					
	2.1.1. Point processes and determinantal correlation functions	374					
	2.1.2. Multivariate OP Ensembles	375					
	2.1.3. The graded lexicographic order and the bijection b	376					
	2.2. A central limit theorem for multivariate OP Ensembles	376					
	2.3. Monte Carlo methods based on determinantal point processes	377					
	2.4. Sampling a multivariate OP Ensemble	380					
3.	Experimental illustration	382					
	3.1. The common setting	382					
	3.2. Crude Monte Carlo: Illustrating Theorem 2.2	383					
	3.3. Importance sampling: Illustrating Theorem 2.3	384					
	3.4. An integrand that violates the assumptions of Theorem 2.2	385					
4.	Discussion and perspectives	385					
Appendix A: Preliminary material							
	A.1. Orthogonal polynomials and the Nevai class	386					

Received March 2017; revised November 2018.

MSC2010 subject classifications. Primary 65C05; secondary 42C05, 60F05.

*Key words and phrases.* Monte Carlo methods, determinantal point processes, hyperuniformity, orthogonal polynomials, central limit theorems.

A.2. Sketch of the proof of Theorem 2.1	90					
A.2.1. Reduction to probability reference measures	90					
A.2.2. Soshnikov's key theorem	90					
A.2.3. Variance asymptotics	91					
A.3. A proof for the upper bound on the limiting variance	91					
A.4. Assumptions of Theorem 2.2 and outline of the proof	92					
A.5. Proof of Theorem 2.4	97					
Appendix B: CLT for multivariate OP Ensembles: Proof of Theorem 2.1	97					
B.1. A useful representation of the covariance	97					
B.2. Covariance asymptotics: The Chebyshev case	98					
Truncated sets and consequences	01					
Proof of (B.16)	01					
Proof of (B.17)	02					
B.3. Covariance asymptotics: The general case	05					
B.4. Extension to $\mathscr{C}^1$ functions and conclusion $\ldots \ldots 4$	08					
Appendix C: Monte Carlo with DPPs: Proof of Theorem 2.2	.09					
C.1. The off-diagonal contribution	10					
C.2. The diagonal contribution	10					
C.3. Proof of Lemma 7	12					
Appendix D: Additional experimental results	13					
Acknowledgements						
References	14					

**1. Introduction.** Numerical integration or quadrature refers to algorithms that approximate integrals

(1.1) 
$$\int f(x)\mu(\mathrm{d}x),$$

where  $\mu$  is a finite positive Borel reference measure, and where f ranges over some class of test functions  $\mathscr{C}$ . We assume for convenience that the support  $\text{Supp}(\mu)$  of  $\mu$  is included in the *d*-dimensional hypercube  $I^d = [-1, 1]^d$ , since one can recover this setting in most applications by means of appropriate transformations. For any given N, a quadrature algorithm outputs N nodes  $\mathbf{x}_1, \ldots, \mathbf{x}_N \in I^d$  and weights  $w_1, \ldots, w_N \in \mathbb{R}$  so that the approximation

(1.2) 
$$\sum_{i=1}^{N} w_i f(\mathbf{x}_i) \approx \int f(x) \mu(\mathrm{d}x)$$

is reasonable for every  $f \in \mathcal{C}$ . The nodes and weights depend on N,  $\mu$ , and can be realizations of random variables, but they are not allowed to depend on f. The quality of a quadrature algorithm is assessed through the approximation error

(1.3) 
$$\mathscr{E}_N(f) = \sum_{i=1}^N w_i f(\mathbf{x}_i) - \int f(x) \mu(\mathrm{d}x)$$

and specifically its behaviour as  $N \to \infty$ .

Many quadrature algorithms have been developed: variations on Riemann summation [18], Gaussian quadrature [27], Monte Carlo methods [67], etc. In the remainder of Section 1, we quickly review three families of such methods to provide context for our contribution, which we then introduce in Section 1.5.

1.1. Gaussian quadrature. Let us first assume d = 1, so that  $\mu$  is supported on I = [-1, 1]. Let  $(\varphi_k)_{k \in \mathbb{N}}$  be the orthonormal polynomials associated with this measure, that is, the family of polynomials such that  $\varphi_k$  has degree k, positive leading coefficient, and  $\int \varphi_k(x)\varphi_\ell(x)\mu(dx) = \delta_{k\ell}$  for every  $k, \ell \in \mathbb{N}$ . Gaussian quadrature (see, e.g., [11, 18, 27] for

general references) then corresponds to taking for nodes  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  the zeros of the *N*th degree orthonormal polynomial  $\varphi_N(x)$ , which are real and simple. As for the weights, Gaussian quadrature corresponds to

(1.4) 
$$w_i = \frac{1}{K_N(\mathbf{x}_i, \mathbf{x}_i)},$$

where we introduced the Nth Christoffel–Darboux kernel associated with  $\mu$ ,

(1.5) 
$$K_N(x, y) = \sum_{k=0}^{N-1} \varphi_k(x) \varphi_k(y).$$

This celebrated method is characterized by the property to be exact, that is,  $\mathscr{E}_N(f) = 0$ , for every polynomial function f of degree up to 2N - 1. This is the highest possible degree such that this holds. Gaussian quadrature is thus particularly suitable when the test functions f look like polynomials. For instance,  $\mathscr{E}_N(f)$  decays exponentially fast when f is analytic [29]. However, although Gaussian quadrature is now two centuries old [25], optimal rates of decay for the error  $\mathscr{E}_N(f)$  do not seem to be known for less regular test functions, say  $f \in \mathscr{C}^1$ , in general. By using Jackson's approximation theorem for algebraic polynomials, one can see that  $\mathscr{E}_N(f) = \mathcal{O}(1/N)$  when  $f \in \mathscr{C}^1$ . Optimal decays have been recently investigated in the particular case of the Gauss–Legendre quadrature [79, 80]. However, even in the familiar Gauss–Jacobi quadrature, optimal rates are only conjectured.

Efficient computation of the nodes and weights in Gaussian quadrature has been an active topic of research. Classical approaches are based on the QR algorithm, such as the Golub–Welsch algorithm; see, for example, [27], Section 3.5, for a discussion. The computational cost of these QR approaches usually scales as  $\mathcal{O}(N^2)$ . More recently,  $\mathcal{O}(N)$  approaches have been proposed for specific choices of the reference measure [30, 33], with parallelizable methods [10] further taking down costs.

Let us stress that Gaussian quadrature is intrinsically a one-dimensional method. Indeed, in the higher-dimensional setting where  $\text{Supp}(\mu) \subset I^d$ , although one may define multivariate orthonormal polynomials associated with  $\mu$ , it is not possible to take for nodes the zeros of a multivariate polynomial. However, if  $\mu$  is a product measure  $\mu = \mu_1 \otimes \cdots \otimes \mu_d$  with each  $\mu_j$ supported on *I*, one could build a grid of nodes using *d* one-dimensional Gaussian quadratures. But this has for consequence to rise up the one-dimensional error estimate for  $\mathcal{E}_N(f)$ to a power 1/d, which essentially makes Gaussian quadrature ineffective in higher dimensions than one or two. In fact, the same phenomenon arises for any other grid-like product of one-dimensional quadratures; this is commonly referred to as the curse of dimensionality.

1.2. Monte Carlo methods. Monte Carlo methods [67] correspond to picking up the N nodes  $(\mathbf{x}_i)$  in (1.2) as the realizations of random variables in  $I^d$ . For instance, assuming  $\mu$  in (1.2) has a density  $\omega$  with respect to the Lebesgue measure, *importance sampling* refers to taking the  $(\mathbf{x}_i)$  to be i.i.d. realizations with a so-called *proposal* density q, and the weights to be

(1.6) 
$$w_i = \frac{1}{N} \frac{\omega(\mathbf{x}_i)}{q(\mathbf{x}_i)}.$$

That way,  $\mathscr{E}_N(f)$  has mean zero. Provided that

(1.7) 
$$\mathbb{V}\mathrm{ar}\bigg[\frac{f(X)\omega(X)}{q(X)}\bigg] < \infty,$$

where X has density q,  $\mathscr{E}_N(f)$  has a standard deviation decreasing as  $N^{-1/2}$ , and satisfies the classical central limit theorem:

$$\sqrt{N}\mathscr{E}_N(f) \xrightarrow[N \to \infty]{\text{law}} \mathcal{N}(0, \sigma_f^2),$$

where  $\sigma_f^2$  equals (1.7). Let us stress that the cost of importance sampling is  $\mathcal{O}(N)$ , and that it can be easily parallelized.

When the ambient dimension d becomes large, practitioners typically prefer Markov chain Monte Carlo (MCMC) methods over importance sampling. This means taking  $w_i = 1/N$  and nodes ( $\mathbf{x}_i$ ) to be the realization of a Markov chain with stationary distribution  $\mu$ , such as the Metropolis–Hastings chain. Under relatively weak conditions on the Markov chain and the integrand,  $\sqrt{N} \mathscr{E}_N(f)$  then converges in distribution to a centered Gaussian variable; see, for example, [23], Theorem 7.32. The limiting variance usually grows more slowly with the dimension d than for importance sampling; see [4] for a proof in a simplified setting. This justifies the preferential use of MCMC for large d. In any case, the typical order of magnitude of the error  $\mathscr{E}_N(f)$  for Monte Carlo methods is  $N^{-1/2}$ , which is often deemed a rather slow decay.

To retain the simplicity of implementation of Monte Carlo and achieve rates faster than  $N^{-1/2}$ , several authors have proposed postprocessing steps. For instance, Delyon and Portier [19] proposed a variant of importance sampling that still takes nodes as independent draws from some proposal density q, but takes weights to be

(1.8) 
$$w_i = \frac{1}{N} \frac{\omega(\mathbf{x}_i)}{\check{q}^{-i}(\mathbf{x}_i)}$$

where  $\check{q}^{-i}$  is the so-called leave-one-out kernel estimator of the density q of the nodes. Perhaps surprisingly, for smooth enough products  $f\omega$  and the right tuning of kernel parameters,  $\sqrt{N}\mathscr{E}_N(f)$  then converges in probability to zero. Exact rates are investigated by Delyon and Portier [19], and a central limit theorem is proven. We further discuss their results in Section 4.

Another postprocessing technique with fast convergence relies on control variates [31]. Oates, Girolami and Chopin [57], for instance, sample nodes i.i.d. from q, and then split the nodes into two batches. The first batch is used to build an approximation  $\hat{f}$  of the integrand f, with the constraint that  $\int \hat{f}(x)\mu(dx)$  is known. The second batch is used to build an importance sampling estimator like (1.6) but targeting the residual  $f - \hat{f}$ . By carefully designing  $\hat{f}$  and controlling the rate at which both batch sizes grow, Oates, Girolami and Chopin [57], Theorem 2, obtain a mean square error in  $N^{-7/6}$  under rather weak assumptions on measure  $\mu$ . The assumptions on the integrand f are stronger, with f to belong to a specific reproducing kernel Hilbert space. We note that Liu and Lee [50] also propose a similar postprocessing approach, but the improvement on the rate is less explicit.

1.3. Quasi-Monte Carlo methods. Quasi-Monte Carlo methods (QMC; [21, 22]) are deterministic constructions that focus on the uniform case,  $\mu(dx) = dx$  in (1.2). The cornerstone of classical QMC is the Koksma–Hlawka inequality [21], equation (3.15). This inequality bounds the error  $\mathscr{E}_N(f)$  in (1.3) by the product of the star discrepancy of the nodes and the Hardy–Krause variation of f. The star discrepancy measures the departure of the empirical measure of the N nodes from the uniform measure. Classical QMC methods aim at proposing efficient node constructions that minimize this star discrepancy. Some constructions guarantee a star discrepancy that asymptotically decreases as fast as  $N^{-1} \log^{d-1} N$ . This implies the same rate for  $\mathscr{E}_N(f)$  provided f has finite Hardy–Krause variation. While this seems faster than typical Monte Carlo methods in Section 1.2, the rate as a function of N does not decrease until N is exponential in d. Moreover, the Hardy–Krause variation is hard to manipulate in practice.

Modern QMC methods come up with more practical rates [21]. For example, scrambled nets [59, 60] are randomized QMC methods, meaning that a stochastic perturbation is applied to a deterministic QMC construction. The perturbation is built so that  $\mathscr{E}_N(f)$  has

mean 0. Owen [59] shows that only assuming f is  $L^2$ , the standard deviation of  $\mathscr{E}_N(f)$  is  $o(N^{-1/2})$ , that is, converges to zero faster than the traditional Monte Carlo rate. When f is smooth enough, which requires at least that all mixed partial derivatives of f of order less than d are continuous, Owen [60] further shows that the standard deviation is  $\mathcal{O}(N^{-3/2-1/d}\log^{(d-1)/2} N)$ . Again, this rate decreases only when N is exponential in the dimension, but Owen [59] shows that for finite N, randomized QMC cannot perform significantly worse than Monte Carlo.

Finally, we note that nonparametric control variates have also been studied for QMC and randomized QMC [56]. While bounds on the error still depend on the rather strong hypotheses of QMC, in particular the smoothness of the integrand, this postprocessing has the advantage of partially bypassing the need for the user to know the degree of smoothness in advance.

1.4. Bayesian quadrature. O'Hagan [55] remarked that if we put a Gaussian process prior [65] over the integrand, then the conditional of its integral given N evaluations is a univariate Gaussian, with a closed-form mean and variance. Picking up nodes by sequentially minimizing this posterior variance yields a range of recent algorithms, such as kernel herding [16] or Bayesian quadrature [36]. There is empirical evidence [36] that the error  $\mathcal{E}_N(f)$  in Bayesian quadrature decreases faster than the Monte Carlo rate  $N^{-1/2}$ . There are theoretical results for hybrid methods between Monte Carlo and Bayesian quadrature [14]; see Section 4 for further discussion.

The nodes and weights of Bayesian quadrature require inverting a  $N \times N$  matrix, and thus the computational cost of the method is  $\mathcal{O}(N^3)$ . Although this cost may seem prohibitive, the approach is justified in some important applications where this cubic computational cost is negligible compared to the cost of one evaluation of the integrand.

1.5. Our contribution. Our main goal is to leverage repulsive particle systems to build a Monte Carlo method with standard deviation of the error decaying as  $o(N^{-1/2})$ . More precisely, the idea is to use correlated random variables for the quadrature nodes, interacting as strongly repulsive particles. Our motivation comes from specific models in random matrix theory (see Section 2.2 for references), for which the linear statistic  $\sum f(x_i)$  converges in distribution to a Gaussian, without requiring any normalizing factor. In this work, we focus on *determinantal point processes* (DPPs), which have received a lot of attention recently in probability and related fields; see, for instance, [35, 39, 46, 49, 51, 74] for a general overview.

In any dimension d, we construct DPPs generating the nodes  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  and appropriate weights  $w_i$ 's so that the error  $\mathscr{E}_N(f)$  in (1.3) decreases rapidly, as  $N \to \infty$ . The general construction of our DPP for an arbitrary  $\mu$  is relatively sophisticated, and will be the topic of Section 2.1. At this stage, we illustrate our results in the specific case where  $\mu$  is the uniform measure on the hypercube  $I^d$ .

THEOREM 1.1. Let  $(P_k)_{k \in \mathbb{N}}$  be the Legendre polynomials defined by recurrence,

$$P_0(x) := 1,$$
  $P_1(x) := x,$   $P_{k+1}(x) := \frac{2k+1}{k+1}xP_k(x) - \frac{k}{k+1}P_{k-1}(x),$ 

and consider, for any  $M \ge 1$  and  $x, y \in \mathbb{R}^d$ , the kernel

$$K_N(x, y) := \prod_{j=1}^d \sum_{k=0}^{M-1} \left(k + \frac{1}{2}\right) P_k(x_j) P_k(y_j),$$

where  $N := M^d$ . Let  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  be sampled with density on  $I^d$ 

(1.9) 
$$\frac{1}{N!} \det \left[ K_N(x_k, x_\ell) \right]_{k,\ell=1}^N$$

Then, for any  $\mathscr{C}^1$  function f that is compactly supported within the open hypercube  $(-1, 1)^d$ ,

$$\mathbb{E}\left[\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)}\right] = \int_{I^d} f(x) \, \mathrm{d}x$$

and the quadratic error satisfies

$$\mathbb{E}\left[\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} - \int_{I^d} f(x) \, \mathrm{d}x\right]^2 = \mathcal{O}\left(\frac{1}{N^{1+1/d}}\right),$$

as  $N \to \infty$ . Moreover, we have a central limit theorem

$$\sqrt{N^{1+1/d}} \left( \sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} - \int_{I^d} f(x) \, \mathrm{d}x \right) \xrightarrow{\mathrm{law}}_{N \to \infty} \mathcal{N}(0, \Omega_{f,1}^2),$$

where  $\Omega_{f,1}$  has an explicit expression involving the regularity of f; see (2.17) below.

Sampling the distribution in (1.9), which is called a multivariate Legendre ensemble, can be done exactly in time  $\mathcal{O}(N^3)$ , if we neglect the cost of rejection sampling steps; see Section 2.4. For the sake of illustration, a sample is shown in Figure 1(a) in the case d = 2. For each *i*, the area of the disk centered at  $\mathbf{x}_i$  is proportional to its weight  $1/K_N(\mathbf{x}_i, \mathbf{x}_i)$  in the quadrature rule of Theorem 1.1. Points are well spread throughout the square, more so than under independent uniform sampling, with a visible accumulation along the border of the square compensated by smaller weights. The green lines on the side plots show the marginals of the uniform measure on the square. The histograms on the side plots are weighted empirical histograms of the sample.

In this paper, we prove a more general result than Theorem 1.1, where we consider a general class of measures instead of the uniform, and relax the assumption that the number of nodes N is a dth power; see Theorem 2.2. For the latter, we need to introduce an ordering on multivariate monomials; see Section 2.2. We also provide an importance sampling (Theorem 2.3) and self-normalized importance sampling (Theorem 2.4) version of this result, having applications to Bayesian inference in mind.

Computationally, our method requires sampling from a DPP. The standard algorithm is  $\mathcal{O}(N^3)$ , plus some overhead due to using rejection sampling routines [35]. As it stands, the applications of Monte Carlo with DPPs are thus naturally the same settings as Section 1.4,



FIG. 1. 1(a) A weighted sample of size N = 150 of the bivariate Legendre ensemble. 1(b) The graded lexicographic order in d = 2: the lower left corner marks the origin  $(0, 0) \in \mathbb{N}^2$ , and the circle with mark k is at coordinates  $\mathfrak{b}(k)$ .

where the bottleneck is the evaluation of the integrand, not the generation of the nodes. Such settings arise in Bayesian inference for simulation-heavy sciences such as astrophysics [78], ecology [64] or cell biology [24], where one evaluation of the integrand requires numerically solving large systems of differential equations. We comment in Section 2.4 on open avenues for faster sampling of specific DPPs, which would further widen the applicability of Monte Carlo with DPPs.

It turns out that, when d = 1, our method is formally very similar to Gaussian quadrature, as described in Section 1.1. We basically replace the zeros of orthogonal polynomials by particles sampled from *orthogonal polynomial ensembles* (OP Ensembles), DPPs whose building blocks are orthogonal polynomials. Our contribution also has the advantage of generalizing more naturally to higher dimensions than Gaussian quadrature through multivariate OP Ensembles.

Monte Carlo with DPPs is to be classified somewhere between classical Monte Carlo methods and QMC methods, respectively, described in Sections 1.2 and 1.3. It is very much similar to importance sampling, but with negatively correlated nodes. Simultaneously, it is more Monte Carlo than scrambled nets, as it does not randomize *a posteriori* a low discrepancy deterministic set of points, but rather incorporate the low discrepancy constraint into the randomization procedure. Our approach also bears similarity with postprocessing approaches to Monte Carlo with fast convergence of Section 1.2. Indeed, the fast convergence in our CLT is due to approximation results, much like nonparametric control variates. We further comment on this in Section 4.

The rest of the paper is organized as follows. In Section 2, we state our quadrature rules and theoretical results on the convergence of its error. In Section 3, we demonstrate our results in simple experimental settings. We conclude with some perspectives in Section 4. In Appendix A, we introduce key technical notions and give the outline of our proofs, the technical parts of the proofs being detailed in Appendices B and C. Appendix D contains additional experimental results.

#### 2. Statement of the results.

NOTATION. All along this work, we write for convenience I = [-1, 1] and  $I^d = [-1, 1]^d$ . Also, for any  $0 < \varepsilon < 1$ , we set  $I_{\varepsilon} = [-1 + \varepsilon, 1 - \varepsilon]$  and  $I_{\varepsilon}^d = [-1 + \varepsilon, 1 - \varepsilon]^d$ . Finally, except when specified otherwise, a *reference measure* is a positive finite Borel measure with support Supp $(\mu) \subset I^d$ .

# 2.1. Determinantal point processes and multivariate OP Ensembles.

2.1.1. Point processes and determinantal correlation functions. A simple point process (hereafter point process) on  $I^d$  is a probability distribution  $\mathbb{P}$  on finite subsets **S** of  $I^d$ . A classical exhaustive reference is [17], but we also refer the reader to [35], which is shorter and contains everything needed in this paper. Given a reference measure  $\mu$ , a point process has a *n*-correlation function  $\rho_n$  if one has

(2.1) 
$$\mathbb{E}\left[\sum_{\substack{\mathbf{x}_{i_1},\dots,\mathbf{x}_{i_n}\in\mathbf{S}\\\mathbf{x}_{i_1}\neq\dots\neq\mathbf{x}_{i_n}}}\varphi(\mathbf{x}_{i_1},\dots,\mathbf{x}_{i_n})\right]$$
$$=\int_{(I^d)^n}\varphi(x_1,\dots,x_n)\rho_n(x_1,\dots,x_n)\mu^{\otimes n}(dx_1,\dots,dx_n)$$

for every bounded Borel function  $\varphi : (I^d)^n \to \mathbb{R}$ , where the sum in (2.1) ranges over all pairwise distinct *n*-uplets of the random finite subset **S**. The function  $\rho_n$ , provided it exists,

thus encodes the correlations between distinct *n*-uplets of the random set **S**. For instance, a Poisson process with intensity  $\lambda : I^d \to \mathbb{R}_+$  is characterized by

(2.2) 
$$\rho_n(x_1,\ldots,x_n) = \prod_{i=1}^n \lambda(x_i)$$

and  $d\mu(x) = \mathbf{1}_{I^d}(x) dx$ . In that particular case, the correlation functions (2.2) are products of univariate terms, which can be paraphrased as *there is no interaction between points in a Poisson point process*. Finally, our definition (2.1) is easily seen to be equivalent to [35], Definition 1, where correlation functions are also called *joint intensities*. For ease of reference, we also note that correlation functions are called *factorial moment densities* in [17], Section 5.4.

A point process is *determinantal* (DPP) if there exists an appropriate kernel  $K : I^d \times I^d \rightarrow \mathbb{R}$  or  $\mathbb{C}$  such that the *n*-correlation function exists for every *n* and reads

(2.3) 
$$\rho_n(x_1, \dots, x_n) = \det \left[ K(x_k, x_\ell) \right]_{k,\ell=1}^n, \quad x_1, \dots, x_n \in I^d.$$

The kernel of a DPP thus encodes how the points in the random configurations interact. The existence of a point process with (2.3) as its correlation functions is, in general, a difficult question. It is easy to see that the kernel has to be positive definite, so that the right-hand side of (2.3) is always nonnegative. But nonnegativity is not sufficient for (2.3) to consistently define a point process.

A canonical way to construct DPPs is to define so-called *projection DPPs*, which generate configurations of N points  $\mathbb{P}$ -almost surely, that is,  $\mathbf{S} = \{\mathbf{x}_1, \dots, \mathbf{x}_N\}$ . More precisely, consider N orthonormal functions  $\varphi_0, \dots, \varphi_{N-1}$  in  $L^2(\mu)$ , that is,  $\int \varphi_k(x)\varphi_\ell(x)\mu(dx) = \delta_{k\ell}$ , and take for kernel

(2.4) 
$$K_N(x, y) = \sum_{k=0}^{N-1} \varphi_k(x) \varphi_k(y).$$

In this setting, it turns out that the (permutation invariant) random variables  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  with joint probability distribution

(2.5) 
$$\frac{1}{N!} \det \left[ K_N(x_i, x_\ell) \right]_{i,\ell=1}^N \prod_{i=1}^N \mu(\mathrm{d} x_i)$$

generate a DPP with kernel  $K_N(x, y)$ . [35], Section 2, gives a proof that (2.5) yields (2.3) with  $K = K_N$ .

For further information on determinantal point processes, we refer the reader to [35, 39, 46, 49, 51, 53, 73].

2.1.2. *Multivariate OP Ensembles.* In the one-dimensional setting, we can for instance build a DPP using (2.5) with  $\varphi_0, \ldots, \varphi_{N-1}$  the N lowest degree orthonormal polynomials associated with the reference measure  $\mu$ . Such DPPs are known as *OP Ensembles* and have been popularized by random matrix theory; see, for example, [43] for an overview.

Our contribution involves a higher-dimensional generalization of OP Ensembles, relying on multivariate orthonormal polynomials, which we now introduce. Given a reference measure  $\mu$ , assume it has well-defined multivariate orthonormal polynomials, meaning that  $\int P^2(x)\mu(dx) > 0$  for every nontrivial polynomial P. This is for instance true if  $\mu(A) > 0$  for some nonempty open set  $A \subset I^d$ . Now choose an ordering for the multi-indices  $(\alpha_1, \ldots, \alpha_d) \in \mathbb{N}^d$ , that is, pick a bijection  $\mathfrak{b} : \mathbb{N} \to \mathbb{N}^d$ . This gives an ordering of the monomial functions  $(x_1, \ldots, x_d) \mapsto x_1^{\alpha_1} \cdots x_d^{\alpha_d}$ , to which one applies the Gram–Schmidt algorithm. This yields a sequence of orthonormal polynomial functions  $(\varphi_k)_{k \in \mathbb{N}}$ , the multivariate orthonormal polynomials. In this work, we use a specific bijection  $\mathfrak{b}$  defined in Section 2.1.3. Equipped with this sequence  $(\varphi_k)_{k \in \mathbb{N}}$  of multivariate orthonormal polynomials, we finally consider for every *N* the DPP associated with the associated kernel (2.4), that we refer to as the *multivariate OP Ensemble* associated with a reference measure  $\mu$ . When d = 1, it reduces to the classical OP Ensemble.

2.1.3. The graded lexicographic order and the bijection  $\mathfrak{b}$ . We consider the bijection  $\mathfrak{b}$  associated with the graded (with respect to the sup norm) alphabetic order on  $\mathbb{N}^d$ . We start with the usual lexicographic order on  $\mathbb{N}^d$ , defined by saying that  $(\alpha_1, \ldots, \alpha_d) <_{\text{lex}}$  $(\beta_1, \ldots, \beta_d)$  if there exists  $j \in \{1, \ldots, d\}$  such that  $\alpha_i = \beta_i$  for every i < j and  $\alpha_j < \beta_j$ . Now we define the graded lexicographic order as follows. We say that  $(\alpha_1, \ldots, \alpha_d) < (\beta_1, \ldots, \beta_d)$  if either max $\{\alpha_1, \ldots, \alpha_d\} < \max\{\beta_1, \ldots, \beta_d\}$  or max $\{\alpha_1, \ldots, \alpha_d\} = \max\{\beta_1, \ldots, \beta_d\}$  and  $(\alpha_1, \ldots, \alpha_d) <_{\text{lex}} (\beta_1, \ldots, \beta_d)$ . Moreover, from now on we specify the bijection  $\mathfrak{b}$  to be the unique bijection  $\mathbb{N} \to \mathbb{N}^d$  increasing for this order. Otherwise put, set  $\mathfrak{b}(0) = (0, \ldots, 0)$  and  $\mathfrak{b}(n) = \min \mathbb{N}^d \setminus \{\mathfrak{b}(0), \ldots, \mathfrak{b}(n-1)\}$  by induction, where the minimum refers to the graded lexicographic order is that, for every  $M \ge 1$ , the set of the first  $M^d$  indices  $\{\mathfrak{b}(0), \ldots, \mathfrak{b}(M^d - 1)\}$  matches the discrete hypercube

(2.6) 
$$\mathcal{C}_M = \{ \boldsymbol{n} \in \mathbb{N}^d : 0 \le n_1, \dots, n_d \le M - 1 \}.$$

The indices between  $\mathfrak{b}(M^d - 1)$  and  $\mathfrak{b}(M^{d+1} - 1)$  then fill the layer  $\mathcal{C}_{M+1} \setminus \mathcal{C}_M$  by following the usual lexicographic order. For better intuition, we illustrate the order  $\mathfrak{b}$  for d = 2 in Figure 1(b); observe how each layer is filled one after the other.

We are now in position to state our first result on multivariate OP Ensembles, which is the cornerstone for the Monte Carlo methods we introduce later in Section 2.3.

2.2. A central limit theorem for multivariate OP Ensembles. Several central limit theorems (CLTs) have been obtained for determinantal point processes and related models in random matrix theory, but only when the random configurations lie in a one- or two-dimensional domain. See, for instance, [1, 2, 7, 12, 13, 20, 37, 38, 40, 44, 47, 48, 61, 63, 66, 71, 74] for a non-exhaustive list. Although DPPs on higher-dimensional supports have attracted attention in complex geometry [5, 6, 8, 9], in statistics [49, 54], and in physics [69, 77], it seems no CLT has been established yet when  $d \ge 3$ .

Our first result for multivariate OP Ensembles is a CLT for  $\mathscr{C}^1$  test functions when the reference measure  $\mu$  is a product of *d* Nevai-class probability measures on *I*. The exact definition of the Nevai class is postponed until Definition 1, but we now give a simple sufficient condition. As a consequence of Denisov–Rakhmanov's theorem (see Theorem A.1), if a measure on *I* has for Lebesgue decomposition  $\mu(dx) = \omega(x) dx + \mu_s$  (where  $\mu_s$  is orthogonal to the Lebesgue measure) with  $\omega(x) > 0$  almost everywhere, then  $\mu$  is Nevai-class. Denote by  $(T_k)_{k \in \mathbb{N}}$  the normalized Chebyshev polynomials, defined on *I* by

$$T_0 = 1,$$
  $T_k(\cos \theta) = \sqrt{2}\cos(k\theta), \quad k \ge 1.$ 

THEOREM 2.1. Let  $\mu$  be a reference measure with  $\text{Supp}(\mu) \subset I^d$ , and assume  $\mu = \mu_1 \otimes \cdots \otimes \mu_d$  where each  $\mu_j$  is Nevai class (see Definition 1). If  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  stands for the associated multivariate OP Ensemble associated with  $\mu$ , then for every  $f \in \mathscr{C}^1(I^d, \mathbb{R})$ , we have

$$\frac{1}{\sqrt{N^{1-1/d}}} \left( \sum_{i=1}^{N} f(\mathbf{x}_i) - \mathbb{E} \left[ \sum_{i=1}^{N} f(\mathbf{x}_i) \right] \right) \xrightarrow[N \to \infty]{\text{law}} \mathcal{N}(0, \sigma_f^2),$$

where

(2.7) 
$$\sigma_f^2 = \frac{1}{2} \sum_{k_1, \dots, k_d=0}^{\infty} (k_1 + \dots + k_d) \hat{f}(k_1, \dots, k_d)^2$$

and

(2.8) 
$$\hat{f}(k_1, \dots, k_d) = \int_{I^d} f(x_1, \dots, x_d) \prod_{j=1}^d T_{k_j}(x_j) \frac{\mathrm{d}x_j}{\pi \sqrt{1 - x_j^2}}.$$

When d = 1, Theorem 2.1 was obtained by [13] (see also [47] for an alternative proof) but the higher-dimensional case  $d \ge 2$  is novel. We shall restrict to  $d \ge 2$  for the proof of the theorem, which is deferred to Section B. Let us now make a few remarks concerning the statement of Theorem 2.1.

REMARK 1. The limiting variance  $\sigma_f^2$  does not depend on the reference measure  $\mu$ .

REMARK 2. By making the change of variables  $x_j = \cos \theta_j$ , we obtain

$$\hat{f}(k_1, \dots, k_d) = \frac{(\sqrt{2})^{|\{j:k_j \neq 0\}|}}{\pi^d} \int_{[0,\pi]^d} f(\cos \theta_1, \dots, \cos \theta_d) \prod_{j=1}^d \cos(k_j \theta_j) \, \mathrm{d}\theta_j,$$

which is, up to a multiplicative factor, a usual Fourier coefficient.

As a side note, we obtain that the limiting variance in Theorem 2.1 is dominated by an explicit integral, that may be of interest to bound  $\sigma_f^2$  in practice.

**PROPOSITION 1.** For any  $f \in \mathscr{C}^1(I^d, \mathbb{R})$ , we have the inequality

(2.9) 
$$\sigma_f^2 \le \frac{1}{2} \sum_{\alpha=1}^d \int_{I^d} \left( \sqrt{1 - x_\alpha^2} \partial_\alpha f(x_1, \dots, x_d) \right)^2 \prod_{j=1}^d \frac{\mathrm{d} x_j}{\pi \sqrt{1 - x_j^2}}.$$

It will appear from the proof we provide in Section A.3 that this inequality is sharp, since equality holds whenever f is a linear combination of monomials  $x_1^{\alpha_1} \cdots x_d^{\alpha_d}$  with  $\alpha_j \in \{0, 1\}$ ; see (A.26).

We now turn to Monte Carlo methods based on Theorem 2.1.

2.3. Monte Carlo methods based on determinantal point processes. Consider a reference measure  $\mu$  with support inside  $I^d$ , having well-defined multivariate orthonormal polynomials (say,  $\mu(A) > 0$  for some open set  $A \subset I^d$ ). Let  $K_N(x, y)$  be the Nth Christoffel–Darboux kernel for the associated multivariate OP Ensemble, namely

(2.10) 
$$K_N(x, y) = \sum_{k=0}^{N-1} \varphi_k(x) \varphi_k(y),$$

where  $(\varphi_k)_{k \in \mathbb{N}}$  is the sequence of multivariate orthonormal polynomials associated with  $\mu$  and the graded lexicographic order; see Section 2.1.3. Our quadrature rule is as follows: take for nodes  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  the random points coming from the multivariate OP Ensemble, namely with joint density (2.5), and for weights  $w_i = 1/K_N(\mathbf{x}_i, \mathbf{x}_i)$ . Thus, for any  $\mu$ -integrable function f, our estimator of  $\int f(x)\mu(dx)$  reads

(2.11) 
$$\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)}.$$

One can readily see by taking n = 1 in (2.1)–(2.4) that the estimator (2.11) is unbiased,

(2.12) 
$$\mathbb{E}\left[\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)}\right] = \int f(x)\mu(dx).$$

REMARK 3. For d = 1, comparing (2.11) to (1.4)–(1.5) yields that our method matches Gaussian quadrature except for the nodes, since we replace the zeros of the univariate orthogonal polynomial  $\varphi_N$  by random points drawn from an OP Ensemble. In fact, this replacement is not aberrant since zeros of orthogonal polynomials and particles of associated OP Ensembles get arbitrarily close with high probability as  $N \to \infty$ ; see [34] for further information and generalizations. Notice however that our quadrature rule has the advantage to make sense in any dimension d.

Our next result is a CLT for (2.11), thus giving a precise rate of decay for the error made in the approximation, provided we make regularity assumptions on  $\mu$  and on the class  $\mathscr{C}$  of test functions f. More precisely, recalling the notation  $I_{\varepsilon}^{d} = [-1 + \varepsilon, 1 - \varepsilon]^{d}$ , we consider

(2.13) 
$$\mathscr{C} = \{ f \in \mathscr{C}^1(I^d, \mathbb{R}) : \operatorname{Supp}(f) \subset I_{\varepsilon}^d \text{ for some } \varepsilon > 0 \}.$$

As for the reference measure, we shall assume  $\mu$  is a product measure with a density which is  $\mathscr{C}^1$  and positive on the open set  $(-1, 1)^d$ . Set for convenience

(2.14) 
$$\omega_{\text{eq}}^{\otimes d}(x) = \prod_{j=1}^{d} \frac{1}{\pi \sqrt{1 - x_j^2}}, \quad x \in I^d.$$

THEOREM 2.2 (Crude Monte Carlo with OP Ensembles). Let  $\mu(dx) = \omega(x) dx$  be a product reference measure with  $\omega(x) = \omega_1(x_1) \cdots \omega_d(x_d)$  and  $\text{Supp}(\mu) \subset I^d$ . Assume  $\omega$  is  $\mathscr{C}^1$  and positive on the open set  $(-1, 1)^d$ , and satisfies: for every  $\varepsilon > 0$ ,

(2.15) 
$$\frac{1}{N} \sup_{x \in I_{\varepsilon}^{d}} |\nabla K_{N}(x, x)| < \infty$$

If  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  stands for the multivariate OP Ensemble associated with  $\mu$ , then for every  $f \in \mathcal{C}$ , we have for the mean square error of the estimator,

(2.16) 
$$\lim_{N \to \infty} N^{1+1/d} \mathbb{E}\left[\left(\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} - \int f(x)\mu(\mathrm{d}x)\right)^2\right] = \Omega_{f,\omega}^2,$$

where (see (2.8))

(2.17) 
$$\Omega_{f,\omega}^2 = \frac{1}{2} \sum_{k_1,\dots,k_d=0}^{\infty} (k_1 + \dots + k_d) \left( \widehat{\frac{f\omega}{\omega_{eq}^{\otimes d}}} \right) (k_1,\dots,k_d)^2.$$

Moreover, we have the central limit theorem,

$$\sqrt{N^{1+1/d}} \left( \sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} - \int f(x) \mu(\mathrm{d}x) \right) \xrightarrow{\mathrm{law}} \mathcal{N}(0, \Omega_{f, \omega}^2).$$

We will discuss the assumptions of Theorem 2.2 in Section A.4 but let us already state that, as it will appear in the proof, (2.15) can be replaced by the weaker but technically involved Assumption 1. We restricted ourselves to (2.15) for the sake of presentation, as it already covers the Jacobi case. Indeed, we prove the following result in Section A.4.

**PROPOSITION 2.** Given any parameters  $\alpha_1, \beta_1, \ldots, \alpha_d, \beta_d > -1$ , the reference measure

(2.18) 
$$\mu(\mathrm{d}x) = \prod_{j=1}^{d} (1-x_j)^{\alpha_j} (1+x_j)^{\beta_j} \mathbf{1}_I(x_j) \,\mathrm{d}x_j,$$

satisfies the assumptions of Theorem 2.2.

Hereafter, we call measures of the form (2.18) Jacobi measures. From a practical point of view, Theorem 2.2 requires knowledge on the measure  $\mu$ , in particular all its moments should be known, since we need the corresponding orthonormal polynomials. This is the case for most applications of Gaussian quadrature, where the reference measure is such that orthonormal polynomials are computable, like Jacobi measures (2.18) for instance. When the moments of  $\mu$  are not known or when  $\mu$  is not separable, we propose an importance sampling result in Theorem 2.3, which shifts most hypotheses onto an instrumental density q. Note however that Theorem 2.3 still requires that we can evaluate the density  $\omega$  of  $\mu$  pointwise.

THEOREM 2.3 (Importance sampling with OP Ensembles). Let  $\mu(dx) = \omega(x) dx$  be a reference measure on  $I^d$  with a  $\mathscr{C}^1$  density  $\omega$  on the open set  $(-1, 1)^d$ . Consider a measure q(x) dx satisfying the assumptions of Theorem 2.2, let  $K_N(x, y)$  be the Nth Christoffel-Darboux kernel associated with q(x) dx, and  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  the associated multivariate OP Ensemble. Then, for every  $f \in \mathscr{C}$ , we have

(2.19) 
$$\mathbb{E}\left[\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} \frac{\omega(\mathbf{x}_i)}{q(\mathbf{x}_i)}\right] = \int f(x)\mu(dx),$$

and we have for the mean square error of the estimator,

(2.20) 
$$\lim_{N \to \infty} N^{1+1/d} \mathbb{E}\left[\left(\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} \frac{\omega(\mathbf{x}_i)}{q(\mathbf{x}_i)} - \int f(x)\mu(\mathrm{d}x)\right)^2\right] = \Omega_{f,\omega}^2,$$

where  $\Omega_{f,\omega}^2$  is the same as (2.17). Moreover, we have the central limit theorem,

(2.21) 
$$\sqrt{N^{1+1/d}} \left( \sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} \frac{\omega(\mathbf{x}_i)}{q(\mathbf{x}_i)} - \int f(x)\mu(\mathrm{d}x) \right) \xrightarrow{\mathrm{law}}_{N \to \infty} \mathcal{N}(0, \Omega_{f, \omega}^2).$$

Indeed, Theorem 2.3 follows from Theorem 2.2 by taking  $f\omega/q$  for test function with  $f \in \mathcal{C}$  and q(x) dx for reference measure.

REMARK 4. From a classical importance sampling perspective, it is surprising that the limiting variance in (2.21) does not depend on the proposal density q.

In most applications to Bayesian inference,  $\mu$  is a probability measure, but its density  $\omega$  can only be evaluated up to a multiplicative constant. A classical trick is to rely on self-normalized importance sampling. Theorem 2.4 states a central limit theorem for such an estimator.

THEOREM 2.4 (Self-normalized importance sampling with OP Ensembles). Let  $\mu(dx) = \omega(x) dx = \omega_u(x) dx/Z$  be a reference probability measure on  $I^d$  with a  $\mathscr{C}^1$  density  $\omega$  on the open set  $(-1, 1)^d$ . We further assume that  $\mu$  is supported in  $I_{\varepsilon}^d = [-1 + \varepsilon, 1 - \varepsilon]^d$ . As in Theorem 2.3, consider a measure q(x) dx satisfying the assumptions of Theorem 2.2,

let  $K_N(x, y)$  be the Nth Christoffel–Darboux kernel associated with q(x) dx, and  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  the associated multivariate OP Ensemble. Finally, for convenience, we let

$$\mathscr{I}_N(g) = \sum_{i=1}^N \frac{g(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} \frac{\omega_u(\mathbf{x}_i)}{q(\mathbf{x}_i)}, \quad g: I^d \to \mathbb{R}$$

Then, for every  $f \in \mathcal{C}$ , we have

(2.22) 
$$\sqrt{N^{1+1/d}} \left( \mathscr{I}_N(f) / \mathscr{I}_N(1) - \int f(x) \mu(\mathrm{d}x) \right) \xrightarrow[N \to \infty]{\text{law}} \mathcal{N}(0, \Xi_{f, \omega}^2),$$

where

$$\Xi_{f,\omega}^2 = \Omega_{f,\omega}^2 - 2c_{f,\omega} \int f \,\mathrm{d}\mu + \Omega_{1,\omega}^2 \left(\int f \,\mathrm{d}\mu\right)^2 \ge 0,$$

and

$$c_{f,\omega} = \frac{1}{2} \sum_{k_1,\dots,k_d=0}^{\infty} (k_1 + \dots + k_d) \left( \frac{\widehat{f\omega}}{\omega_{eq}^{\otimes d}} \right) (k_1,\dots,k_d) \times \left( \frac{\widehat{\omega}}{\omega_{eq}^{\otimes d}} \right) (k_1,\dots,k_d).$$

We prove Theorem 2.4 in Section A.5 using the same arguments as for classical importance sampling [70], Section 7.1.3, but replacing the standard CLT by Theorem 2.3.

2.4. Sampling a multivariate OP Ensemble. For Monte Carlo with DPPs to be a practical tool, we need to be able to sample realizations of the random variables  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  with joint density (2.5). [35] give an algorithm for sampling generic DPPs, which we use here; see also [49, 58, 69] for more details. In terms of code, a companion Python package to the current paper is available,<sup>1</sup> which implements the OPE sampling described in this section and used later for the experiments in Section 3. A more efficient implementation of the same OPE sampling algorithm, along with most known DPP sampling algorithms, can also be found in the Python package DPPy<sup>2</sup> 26.

The algorithm is based on the fact that the chain rule for the joint distribution (2.5) is available as

$$\frac{1}{N!} \det[K_N(x_i, x_\ell)]_{i,\ell=1}^N \prod_{i=1}^N \mu(dx_i)$$

(2.23)

$$=\prod_{i=1}^{N}\frac{1}{N-i+1}\|P_{H_{i-1}}K_N(x_i,\cdot)\|_{L^2(\mu)}^2\mu(\mathrm{d} x_i).$$

In (2.23),  $P_H$  is the orthogonal projection onto a subspace H of  $L^2(\mu)$ ,

 $H_0 = \operatorname{Span}(\varphi_0, \ldots, \varphi_{N-1}),$ 

and  $H_{i-1}$  is the orthocomplement in  $H_0$  of

$$\operatorname{Span}(K_N(x_\ell, \cdot), 1 \le \ell \le i - 1)$$

for every i > 1. In particular, all the terms in the product of the RHS of (2.23) are probability measures [35], Proposition 19. Notice that the factorization (2.23) is the equivalent of the "base times height" formula that computes the squared volume of the parallelotope generated by the vectors ( $\varphi_0(x_i), \ldots, \varphi_{N-1}(x_i)$ ) for  $1 \le i \le N$ .

<sup>&</sup>lt;sup>1</sup>https://github.com/rbardenet/dppmc

<sup>&</sup>lt;sup>2</sup>https://github.com/guilgautier/DPPy

Using the normal equations, we can also rewrite each term in the product (2.23)

(2.24)  
$$\|P_{H_{i-1}}K_N(x_i, \cdot)\|_{L^2(\mu)}^2 = \begin{cases} K_N(x_1, x_1) & \text{if } i = 1, \\ K_N(x_i, x_i) - \mathbf{k}_{1:i-1}(x_i)^T \mathbf{K}_{1:i-1}^{-1} \mathbf{k}_{1:i-1}(x_i) & \text{else,} \end{cases}$$

where

$$\mathbf{k}_{1:i-1}(\cdot) = \left(K_N(x_1, \cdot), \dots, K_N(x_{i-1}, \cdot)\right)^T$$

and

$$\mathbf{K}_{1:i-1} = \left[ K_N(x_k, x_\ell) \right]_{1 \le k, \ell \le i-1}$$

REMARK 5. Equation (2.24) will be familiar to users of Gaussian processes (GPs; Rasmussen and Williams [65]): the unnormalized conditional densities (2.24) are the incremental posterior variances in a GP model with the same kernel.

REMARK 6. Evaluating (2.24) requires evaluating  $K_N$ , or equivalently the polynomials  $\varphi_k$  for k = 0, ..., N - 1. This can be efficiently implemented using the three-term recurrence relations for orthogonal polynomials, when the recurrence coefficients are known; see, for example, [27], Section 1.3, for whom this recurrence is "arguably the single most important piece of information for the constructive and computational use of orthogonal polynomials."

In a nutshell, sampling a multivariate OPE amounts to sampling from each conditional (2.24) in the chain rule (2.23), one after the other. The only thing left to specify is how we sample each conditional. In this paper, we propose to sample each conditional by rejection sampling [67], Section 2.3. This requires proposal densities  $(q_i)_{1 \le i \le N}$  and tight bounds on the density ratios

(2.25) 
$$\frac{\|P_{H_{i-1}}K_N(x,\cdot)\|_{L^2(\mu)}^2\omega(x)}{q_i(x)}, \quad 1 \le i \le N,$$

when  $\mu(dx) = \omega(x) dx$ . A theorem of Totik, which we recall later as Theorem A.4, gives light conditions on  $\omega$ , under which

$$\frac{N}{K_N(x,x)} \to \frac{\omega(x)}{\omega_{\rm eq}(x)}$$

uniformly on  $I_{\varepsilon}^{d}$ . This suggests choosing

$$q_i(x) = q(x) = \omega_{\text{eq}}^{\otimes d}(x) = \prod_{j=1}^d \frac{1}{\pi \sqrt{1 - x_j^2}} \mathbf{1}_{[-1,1]}(x_j), \quad 1 \le i \le N$$

To bound (2.25), it is enough to bound  $K_N(x, x)\omega(x)/\omega_{eq}^{\otimes d}(x)$  since  $K_N$  is a positive definite kernel. Obtaining tight bounds is problem-dependent. Interestingly, for Jacobi measures, these bounds have been an active topic of research and we can use, for example, the bounds in [28] for our rejection sampling. This means that in practice, we can apply our method in the classical cases where Gaussian quadrature is applied.

We now discuss the cost of sampling a multivariate OP Ensemble. Without taking into account the evaluation of orthogonal polynomials nor rejection sampling,<sup>3</sup> the number of basic

 $<sup>^{3}</sup>$ The cost of the rejection steps, in particular, depends on the tightness of the bound of (2.25) and would need further study.

operations is as much as for Gram–Schmidt orthogonalization of N vectors of dimension N, that is of order  $N^3$  [32], Section 5.2. This means that Monte Carlo with DPPs is to be used when the gain in accuracy in Theorem 2.3 is worth spending a cubic computational budget to obtain the quadrature nodes. Such settings arise in Bayesian inference with expensive models in the natural sciences, where evaluating the integrand once can easily be a question of hours or more. Those are the same application areas as discussed for Bayesian quadrature in Section 1.4.

Additionally, we note that our central limit Theorems 2.2, 2.3 and 2.4 are independent of the algorithm we use to sample the multivariate OP Ensemble. Should a faster algorithm come out, this would further augment the applicability of Monte Carlo with DPPs. Fast sampling algorithms are out of the scope of this paper, but there are reasons to think they do exist. First, when d = 1 and the reference measure is Jacobi (2.18), sampling the OP Ensemble can already be done rejection-free and in time  $O(N^2)$  by diagonalizing a tridiagonal random matrix that only requires sampling independent beta variables [42]. Second, some discrete examples of DPPs can also be sampled in time  $O(N \log N)$  [52], Chapter 4. Third, since Monte Carlo with DPPs closely connects with methods such as QMC (Section 1.3) and Bayesian quadrature (Section 1.4), inspiration could be drawn from fast methods that exist for these families of algorithms [3, 15, 21].

**3. Experimental illustration.** In this section, we illustrate Theorems 2.2 and 2.3 with three toy experiments. In particular, for both CLTs, we investigate how fast the Gaussian limit appears in each theorem and we estimate the rate of decay of the variance.

3.1. The common setting. We consider OP Ensembles with reference measure the product Jacobi measure (2.18) with  $\alpha_1 = \beta_1 = -1/2$ , and  $\alpha_j$ ,  $\beta_j$  drawn i.i.d. uniformly on [-1/2, 1/2] for  $1 < j \le d$ . As proposed in Section 2.4, we use  $\omega_{eq}^{\otimes d}$  for the density of the proposal in the rejection sampling steps, and the bounds in [28]. For various  $N \in [10, 400]$ and each dimension  $d \in \{1, 2, 3\}$ , we sample  $N_{repeat} = 100$  independent realizations of  $\{\mathbf{x}_1, \ldots, \mathbf{x}_N\}$ . We refer to Section 2.4 for details on the sampling algorithm and references to implementations.

Figure 2(a) depicts one of the obtained samples when d = 2 and N = 150. Each disk is centered at a node  $\mathbf{x}_i$  in the sample, and the area of the disk is proportional to the weight  $1/K_N(\mathbf{x}_i, \mathbf{x}_i)$ . The marginal plots on each axis depict the marginal histograms of the weighted



FIG. 2. 2(a) A weighted sample of the Jacobi OPE described in Section 3.1. 2(b) The test function (3.1) when d = 1, along with the proposal and the target used for demonstrating the importance sampling procedure in Theorem 2.3.

sample, with a green curve indicating the density of the marginal Jacobi measures corresponding to j = 1, 2 in (2.18). Good agreement is observed for the marginals, as expected from the unbiasedness in (2.12).

We now proceed to investigating the Gaussianity and estimating the variance decay of the linear statistics in Theorems 2.2 and 2.3, with three integration tasks. All our confidence intervals include a Bonferroni correction to take into account the fact that these three experiments share the same OPE samples.

3.2. *Crude Monte Carlo: Illustrating Theorem* 2.2. We define a simple "bump" test function that is  $\mathscr{C}^{\infty}$  on  $I^d = [-1, 1]^d$  and vanishes outside  $I_{\varepsilon}^d = [-1 + \varepsilon, 1 - \varepsilon]^d$ ,

(3.1) 
$$f(x) = \mathbf{1}_{I_{\varepsilon}^{d}}(x) \prod_{j=1}^{d} \exp\left(-\frac{1}{1-\varepsilon - x_{j}^{2}}\right),$$

so that  $f \in \mathscr{C}$  and thus satisfies the assumptions of Theorem 2.2. We set  $\varepsilon = 0.05$  and plot f for d = 2 in Figure 3(a).

We summarize our results for each dimension d in Figure 4. On each quadrant and for each N, we plot in black circles the sample variance of

$$\sum_{i=1}^{N} \frac{f(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)},$$

computed over the  $N_{\text{repeat}}$  realizations. Blue and red dots indicate standard confidence intervals, for indication only. For comparison, we also plot in white circles the sample variance of a standard crude Monte Carlo estimator of the same integral, using i.i.d. samples of the reference measure.

For a given dimension d, we want to infer the rate of decay of the variance, in order to confirm the rate in the CLT of Theorem 2.2. We proceed as follows. We first select the values of N for which the  $N_{\text{repeat}}$  realizations give a p-value larger than 0.05 in a Kolmogorov–Smirnov test of Gaussianity. This is meant to eliminate the small values of N for which the Gaussian in the CLT (2.21) is a bad approximation for our samples. We do not claim to perform any multiple testing, but rather use the p-value as a loose indicator of Gaussianity. The bottom plot of each quadrant of Figure 4 shows the p-values as a function of N. Note how Gaussianity is hinted even for small N in d = 1, 2, while for d = 3, it takes larger N to kick in. Then we perform a standard frequentist linear regression on the selected log variances versus



FIG. 3. The two test functions.



FIG. 4. Summary of the crude Monte Carlo results.

log(N). For visualization, we plot on each quadrant of Figure 4 the maximum likelihood (MLE) line in green and indicate its slope in the legend. The usual Student-t confidence intervals for the slope are given in the column of Table 1 labeled *Crude MC*.

The confidence intervals are in very good agreement with Theorem 2.2 for each dimension d. The combined plots in Figure 4 hint that the CLT approximation is strikingly accurate for all d, even for small N. For d = 3, the Gaussianity appears slightly later in terms of N, which confirms the intuition that the convergence to a Gaussian is slower when the dimension increases. Relatedly, the intercept of the various straight lines increases with d, and this increase seems to be faster for DPPs than crude Monte Carlo. This entails that the value of N above which OPEs become significantly more efficient than crude Monte Carlo increases with d.

3.3. Importance sampling: Illustrating Theorem 2.3. We now illustrate the importance sampling result in Theorem 2.3. As proposal reference measure, we use the OPEs described in Section 3.1. More precisely, we take q(x) dx in Theorem 2.3 to be the product Jacobi measure described in Section 3.1. The goal is still to estimate the integral of  $\varphi$  in (3.1), but

d	-1 - 1/d	Crude MC	Importance sampling	Assumption violation
1	-2	[-2.05, -1.92]	[-2.13, -1.99]	[-2.08, -1.92]
2	-1.5	[-1.55, -1.43]	[-1.61, -1.44]	[-1.60, -1.42]
3	-1.33	[-1.49, -1.33]	[-1.42, -1.16]	[-1.24, -1.07]

TABLE 1

Confidence intervals for the variance decay, for all three experiments of Section 3. In bold, we highlight the only confidence interval that does not contain the corresponding theoretical rate given in the second column

this time with respect to a target distribution  $\mu$  that is a truncated mixture of two Gaussians, with density

$$\omega(x) = \frac{1}{2} \exp\left[-\frac{\|x - 0.5\|^2}{(0.3)^2}\right] + \exp\left[-\frac{\|x + 1\|^2}{(0.5)^2}\right]$$

with respect to the Lebesgue measure on  $I^d$ . The target measure  $\mu$ , the proposal reference measure and the test function are illustrated in d = 1 in Figure 2(b).

For the sake of shortness, we defer the figure displaying the results of the regression to Figure 5 in Appendix D. We only report here the resulting confidence intervals on the rate of decay of the variance, in the column of Table 1 labeled *importance sampling*. Again, the confidence intervals are in very good agreement with the CLT in Theorem 2.3.

3.4. An integrand that violates the assumptions of Theorem 2.2. We again copy the setting of Section 3.1, but we change the integrand to

$$f(x) = d^{-1} \|x\|_1.$$

This test function is plotted in Figure 3(b) for d = 2. It does not belong to the class  $\mathscr{C}$  that we authorize in Theorem 2.2: it is not  $\mathscr{C}^1$ , and it does not vanish at the border of  $I^d$ . Again, we defer the display of the regression to Figure 6 in Appendix D, and we limit ourselves here to the confidence intervals on the slope, which are given in the last column of Table 1. This time, while the decay of the mean square error is still significantly better than crude Monte Carlo, the confidence intervals do not all match the conclusion of Theorem 2.2: we have written in bold the confidence interval for d = 3, which suggests a rate of decay that is slower than -1 - 1/d. This is a hint that the assumptions of Theorem 2.2 are essentially tight.

4. Discussion and perspectives. As detailed in Remark 3, Monte Carlo with DPPs is a stochastic counterpart to Gaussian quadrature, introduced in Section 1.1. Compared to the Monte Carlo methods introduced in Section 1.2, and 1.3, Theorem 2.3 is an importance sampling procedure, with negatively correlated importance samples. This negative correlation results in a variance reduction that impacts the decay rate of the variance. Loosely speaking, this is reminiscent of the surprising kernel density approach to importance sampling of [19] described in Section 1.2. Our rates are better for equivalent smoothness in d = 1, but for d > 1, the theoretical comparison is less clear. In terms of sampling cost, [19] scales as  $\mathcal{O}(N^2)$  not taking into account the tuning of the kernel parameters. Naively sampling orthogonal polynomial ensembles is  $\mathcal{O}(N^3)$ , without taking rejection sampling into account. Tackling the cubic cost of sampling DPPs is a natural sequel to our work; see also Section 2.4.

Monte Carlo with DPPs is also reminiscent of randomized quasi-Monte Carlo methods such as scrambled nets [59], discussed in Section 1.3. The important difference is that randomness and discrepancy are tied in our DPP proposal. The similarities with QMC are an interesting lead for future research. In particular, fast constructions of nets in QMC [21] could yield fast sampling algorithms for DPPs.

Monte Carlo with DPPs also connects with Bayesian quadrature, introduced in Section 1.4. As pointed out in Section 2.4, sampling projection DPPs is related to sequentially maximizing the variance of a Gaussian process, while Bayesian quadrature is about sequentially minimizing the variance of the integral of f when a Gaussian process prior is assumed on f; see Section 1.4. A formal connection with Bayesian quadrature would facilitate the transfer of CLTs such as our Theorem 2.3 to Bayesian quadrature. Conversely, the efficient Frank–Wolfe optimization procedures given for herding [3, 15] could influence fast sampling algorithms for DPPs.

Additionally, theoretical rates have been provided for hybrid integral estimators [14], Theorem 1, which mix classical Monte Carlo nodes with Bayesian quadrature weights, effectively reducing the influence of close-by pairs of Monte Carlo samples. Together with [19], the latter approach uses a kernel to introduce anti-correlation in a *postprocessing* reweighting step. In comparison, Monte Carlo with DPPs also uses a kernel to encode repulsiveness, but the repulsiveness not only appears in the weights: it is embedded in the sampling procedure.

The same reason that sampling and approximation are both encapsulated in the same DPP object is the major difference with the postprocessing approaches of [50, 57]. With DPPs, we lose the algorithmic simplicity and often low cost of postprocessing, but we gain a CLT with weaker, *dimension-independent* smoothness requirements. Also, our importance reweighted estimator in Theorem 2.3 bypasses the need to know the moments of the target measure, which is akin to removing the constraint of knowing the integral of the control approximation to the integrand in [57]. There are interesting avenues to try to obtain a hybrid algorithm that would get the best of both worlds.

Finally, we comment on our focus on orthogonal polynomials and projection kernels. Relying on orthogonal polynomials made available technology that we extensively used in the proofs, such as precise asymptotic results and the use of recurrence coefficients. First, with slightly stronger estimates, one may allow the reference measure to depend on N, so as to replace the equilibrium measure  $\mu_{eq}^{\otimes d}$  by a measure putting less mass on the boundary of the integration domain. This would prevent part of the quadrature nodes to clutter close to the boundary of the hypercube. Second, any projection kernel onto an N-dimensional subspace of  $L^2(\mu)$  yields an appropriate DPP for numerical integration, and orthogonal polynomials may not be the most natural choice of basis for a given integrand. It is easy to imagine kernels built on wavelets or other bases of  $L^2(\mu)$ , and a clever choice of basis may also yield faster sampling algorithms, but the difficulty lies in obtaining such variance estimates as we obtained for orthogonal polynomials. Third, one should keep in mind that not every projection kernel leads to small variance. For instance, the projection kernel onto

$$\operatorname{Span}(e^{2ik\pi\theta}:k=0,\ldots,N-1)$$

yields a DPP on [-1, 1] such that  $\mathbb{V}ar[\sum f(\mathbf{x}_i)] \sim N$  for any odd function of  $L^2(-1, 1)$ . On the positive side, extra complex structure can buy us further variance reduction in even dimensions for analytic test functions. For instance, one can show along the lines of Section B.2, that the projection onto  $\operatorname{Span}(z^k : k = 0, ..., N - 1)$  yields a DPP on the unit disc with  $\mathbb{V}ar[\sum f(\mathbf{x}_i)] \sim 1$  instead of  $N^{1-1/2}$  provided the function f is analytic. Fourth, one may be tempted to use contraction kernels instead of projection kernels, but besides the problem that the number of points drawn from the DPP becomes random, contraction DPPs are bound to augment the variance of linear statistics.

To conclude, DPPs are a new way to connect numerical integration with rich analytic tools.

### APPENDIX A: PRELIMINARY MATERIAL

In this section, we provide some general background on orthogonal polynomials, we prove short results and we outline the proofs of the main theorems.

A.1. Orthogonal polynomials and the Nevai class. In the following, we use the equilibrium measure  $\mu_{eq}$  of *I*, defined by

(A.1) 
$$\mu_{eq}(dx) = \omega_{eq}(x) dx, \qquad \omega_{eq}(x) = \frac{1}{\pi \sqrt{1 - x^2}} \mathbf{1}_I(x).$$

The name comes from its characterization as the unique minimizer of the logarithmic energy  $\iint \log |x-y|^{-1} \mu(dx) \mu(dy)$  over Borel probability measures  $\mu$  on *I* [68]. It is also the image

of the uniform measure on the unit circle through the map  $e^{i\theta} \mapsto x = \cos\theta$ . The associated orthonormal polynomials are the normalized Chebyshev polynomials of the first kind, defined on *I* by

(A.2) 
$$T_k(\cos\theta) = \begin{cases} \sqrt{2}\cos(k\theta) & \text{if } k \ge 1, \\ 1 & \text{if } k = 0, \end{cases} \quad \theta \in [0, \pi].$$

They satisfy the three-term recurrence relation

(A.3) 
$$xT_k(x) = a_k^* T_{k+1}(x) + b_k^* T_k(x) + a_{k-1}^* T_{k-1}(x), \quad k \in \mathbb{N},$$

where

(A.4) 
$$a_k^* = \begin{cases} 0 & \text{if } k = -1, \\ 1/\sqrt{2} & \text{if } k = 0, \\ 1/2 & \text{if } k \ge 1 \end{cases} \text{ and } b_k^* = 0.$$

More generally, given a reference measure  $\mu$  on I with orthonormal polynomials ( $\varphi_k$ ), we always have the three-term recurrence relation

(A.5) 
$$x\varphi_k(x) = a_k\varphi_{k+1}(x) + b_k\varphi_k(x) + a_{k-1}\varphi_{k-1}(x), \quad k \in \mathbb{N},$$

where  $a_{-1} = 0$  and  $a_k > 0$  and  $b_k \in \mathbb{R}$  for every  $k \ge 0$ . The existence of the recurrence coefficients  $(a_k)_{k\in\mathbb{N}}$  and  $(b_k)_{k\in\mathbb{N}}$  follows by decomposing the polynomial  $x\varphi_k$  into the orthonormal family  $(\varphi_\ell)_{\ell=0}^{k+1}$  of  $L^2(\mu)$  and observing that  $\langle x\varphi_k, \varphi_\ell \rangle = \langle \varphi_k, x\varphi_\ell \rangle = 0$  as soon as  $\ell < k-1$  by orthogonality.

DEFINITION 1. A measure  $\mu$  supported on *I* is *Nevai-class* if the recurrence coefficients for the associated orthonormal polynomials satisfy

$$\lim_{k \to \infty} a_k = 1/2, \qquad \lim_{k \to \infty} b_k = 0.$$

Notice the respective limits of the  $a_k$ 's and  $b_k$ 's for Nevai class measures are the recurrence coefficients (A.4) of the measure  $\mu_{eq}$  when  $k \ge 1$ .

The next theorem gives a sufficient condition for a measure to be Nevai class [72], Theorem 1.4.2.

THEOREM A.1 (Denisov–Rakhmanov). Let  $\mu$  be a reference measure on I with Lebesgue decomposition  $\mu(dx) = \omega(x) dx + \mu_s$ . If  $\omega(x) > 0$  almost everywhere, then  $\mu$  is Nevai-class.

Consider now the Christoffel-Darboux kernel

(A.6) 
$$K_N(x, y) = \sum_{k=0}^{N-1} \varphi_k(x)\varphi_k(y),$$

and notice  $\frac{1}{N}K_N(x, x)\mu(dx)$  is a probability measure. One of the interesting properties of Nevai-class measures is that this probability measure has  $\mu_{eq}$  for weak limit as  $N \to \infty$  [75].

THEOREM A.2. Assume  $\mu$  supported on I is Nevai-class. Then, for every  $f \in \mathscr{C}^0(I, \mathbb{R})$ ,

$$\int f(x) \frac{1}{N} K_N(x, x) \mu(\mathrm{d} x) \xrightarrow[N \to \infty]{} \int f(x) \mu_{\mathrm{eq}}(\mathrm{d} x).$$

Now, consider instead a reference measure  $\mu$  on  $I^d$  with associated multivariate orthogonal polynomials  $(\varphi_k)_{k \in \mathbb{N}}$  (see Section 2.1) and Christoffel–Darboux kernel  $K_N(x, y)$  defined as in (A.6). Assume further that  $\mu = \mu_1 \otimes \cdots \otimes \mu_d$  is a product of d measures  $\mu_j$  on I, and denote by  $\varphi_k^{(j)}$  and  $K_N^{(j)}(x, y)$  the respective orthogonal polynomials and Christoffel–Darboux kernel associated with  $\mu_j$ . Then we have

(A.7) 
$$\varphi_k(x) = \varphi_{k_1}^{(1)}(x_1) \cdots \varphi_{k_d}^{(d)}(x_d)$$

where  $(k_1, \ldots, k_d) = \mathfrak{b}(k)$ . Moreover,

(A.8) 
$$K_{M^d}(x, y) = \prod_{j=1}^d K_M^{(j)}(x_j, y_j).$$

As a consequence, Theorem A.2 easily yields the following.

COROLLARY 1. Let  $\mu = \mu_1 \otimes \cdots \otimes \mu_d$  with  $\mu_j$  supported on I and Nevai-class. Then, for every  $f \in \mathscr{C}^0(I^d, \mathbb{R})$ ,

(A.9) 
$$\int f(x) \frac{1}{N} K_N(x, x) \mu(\mathrm{d}x) \xrightarrow[N \to \infty]{} \int f(x) \mu_{\mathrm{eq}}^{\otimes d}(\mathrm{d}x).$$

PROOF. By the Stone–Weierstrass theorem, it is enough to show (A.9) when  $f(x) = \prod_{j=1}^{d} f_j(x_j)$  with  $f_j \in \mathscr{C}^0(I, \mathbb{R})$ . Without loss of generality, one can further assume the functions  $f_j$  are non-negative. Let  $M = \lfloor N^{1/d} \rfloor$  be the unique integer satisfying  $M^d \le N < (M+1)^d$ . Since we have  $K_{M^d}(x, x) \le K_N(x, x) \le K_{(M+1)^d}(x, x)$  and, by (A.8),

$$\begin{split} \frac{M^d}{N} &\prod_{j=1}^d \int f_j(x) \frac{1}{M} K_M^{(j)}(x,x) \mu_j(\mathrm{d}x) \\ &\leq \int f(x) \frac{1}{N} K_N(x,x) \mu(\mathrm{d}x) \\ &\leq \frac{(M+1)^d}{N} \prod_{j=1}^d \int f_j(x) \frac{1}{M+1} K_{M+1}^{(j)}(x,x) \mu_j(\mathrm{d}x), \end{split}$$

Corollary 1 follows from Theorem A.2.  $\Box$ 

The next lemma is yet another aspect of Nevai-class measures that is relevant to our proofs, and may be of independent interest.

LEMMA 1. Assume  $\mu$  supported on I is Nevai-class. We have the weak convergence of

(A.11) 
$$Q_N(dx, dy) = (x - y)^2 K_N(x, y)^2 \mu(dx) \mu(dy)$$

towards

(A.10)

(A.12) 
$$L(dx, dy) = \frac{1}{2}(1 - xy)\mu_{eq}(dx)\mu_{eq}(dy).$$

PROOF. First, the Christoffel–Darboux formula reads

(A.13) 
$$(x - y)^2 K_N(x, y)^2 = a_N^2 (\varphi_N(x)\varphi_{N-1}(y) - \varphi_{N-1}(x)\varphi_N(y))^2,$$

which follows by computing  $xK_N(x, y)$  using the recurrence relation (A.5) and witnessing several cancellations when subtracting  $yK_N(x, y)$ . Thus, by the orthonormality conditions, we see  $\iint Q_N(dx, dy) = 2a_N^2$ . Since  $\mu$  is Nevai-class, the former converges to  $1/2 = \iint L(dx, dy)$ . This allows us to use the usual weak topology (i.e., the topology coming by duality with respect to the continuous functions) for bounded Borel measures.

Step 1. We first prove the lemma when  $\mu = \mu_{eq}$ , so that the  $\varphi_k$ 's are the Chebyshev polynomials  $T_k$ ; see (A.2). By (A.13), the push-forward of (A.11) by the map  $(x, y) \mapsto (\cos \theta, \cos \eta)$ , where  $\theta, \eta \in [0, \pi]$ , reads

(A.14) 
$$\frac{1}{\pi^2} \{\cos(N\theta)\cos((N-1)\eta) - \cos((N-1)\theta\cos(N\eta))\}^2 d\theta d\eta.$$

This measure has for Fourier transform

$$\frac{1}{\pi^2} \int_0^{\pi} \int_0^{\pi} e^{i(\theta u + \eta v)} \{\cos(N\theta) \cos((N-1)\eta) - \cos((N-1)\theta) \cos(N\eta)\}^2 d\theta d\eta$$
$$= \frac{1}{\pi^2} \int_0^{\pi} \int_0^{\pi} \cos(\theta u + \eta v) \{\cos(N\theta) \cos((N-1)\eta) - \cos((N-1)\theta) \cos(N\eta)\}^2 d\theta d\eta.$$

By developing the square in the integrand and linearizing the products of cosines, we see that the nonvanishing contribution as  $N \to \infty$  of the Fourier transform are the terms which are independent on N. Indeed, the N-dependent terms come up with a factor 1/N after integration. Thus, the Fourier transform equals, up to O(1/N), to

$$\frac{1}{2\pi^2} \int_0^{\pi} \int_0^{\pi} \cos(\theta u + \eta v) (1 - \cos\theta \cos\eta) \,\mathrm{d}\theta \,\mathrm{d}\eta.$$

This yields the weak convergence of (A.14) towards  $(2\pi^2)^{-1}(1 - \cos\theta \cos\eta) \,d\theta \,d\eta$ , and the lemma follows, in the case where  $\mu = \mu_{eq}$ , by taking the image of the measures by the inverse map  $(\cos\theta, \cos\eta) \mapsto (x, y)$ .

Step 2. We now prove the lemma for a general Nevai-class measure  $\mu$  on *I*. Let us denote by  $Q_N^{\mu}$  the measure (A.11) in order to stress the dependence on  $\mu$ . Thanks to Step 1, it is enough to prove that for every  $m, n \in \mathbb{N}$ , we have

$$\lim_{N \to \infty} \left| \iint x^m y^n Q_N^{\mu}(\mathrm{d}x, \mathrm{d}y) - \iint x^m y^n Q_N^{\mu_{\mathrm{eq}}}(\mathrm{d}x, \mathrm{d}y) \right| = 0.$$

in order to complete the proof of the lemma. Recalling (A.11), (A.13), and that  $a_N \rightarrow 1/2$ , it is enough to show that for every  $m \in \mathbb{N}$ ,

(A.15) 
$$\lim_{N \to \infty} \left| \int x^m \varphi_N^2(x) \mu(\mathrm{d}x) - \int x^m T_N^2(x) \mu_{\mathrm{eq}}(\mathrm{d}x) \right| = 0$$

and

(A.16) 
$$\lim_{N \to \infty} \left| \int x^m \varphi_N(x) \varphi_{N-1}(x) \mu(\mathrm{d}x) - \int x^m T_N(x) T_{N-1}(x) \mu_{\mathrm{eq}}(\mathrm{d}x) \right| = 0.$$

To do so, we first complete for convenience the sequences of recurrence coefficients  $(a_n)_{n \in \mathbb{N}}$ and  $(b_n)_{n \in \mathbb{N}}$  introduced in (A.5) as bi-infinite sequences  $(a_n)_{n \in \mathbb{Z}}$ ,  $(b_n)_{n \in \mathbb{Z}}$ , where we set  $a_n = b_n = 0$  for every n < 0. It follows inductively from the three-term recurrence relation (A.5) that for every  $k, \ell, m \in \mathbb{N}$ ,

(A.17) 
$$\int x^m \varphi_k(x) \varphi_\ell(x) \mu(\mathrm{d}x) = \sum_{\gamma:(0,k)\to(m,\ell)} \prod_{e\in\gamma} \omega(e)_{\{(a_n),(b_n)\}},$$

where the sum ranges over all the paths  $\gamma$  lying on the oriented graph with vertices  $\mathbb{Z}^2$  and edges  $(i, j) \rightarrow (i + 1, j + 1), (i, j) \rightarrow (i + 1, j)$  and  $(i, j) \rightarrow (i + 1, j - 1)$  for  $(i, j) \in \mathbb{Z}^2$ ,

starting from (0, k) and ending at  $(m, \ell)$ . For every edge *e* of  $\mathbb{Z}^2$ , we introduced the weight associated with the sequences  $(a_n) = (a_n)_{n \in \mathbb{Z}}$ ,  $(b_n) = (b_n)_{n \in \mathbb{Z}}$  defined by

(A.18) 
$$\omega(e)_{\{(a_n),(b_n)\}} = \begin{cases} a_j & \text{if } e = (i, j) \to (i+1, j+1), \\ b_j & \text{if } e = (i, j) \to (i+1, j), \\ a_{j-1} & \text{if } e = (i, j) \to (i+1, j-1); \end{cases}$$

see also [34]. Now, observe that the set of all paths  $\gamma$  satisfying  $\gamma : (0, k) \to (m, \ell)$  only depends on  $k, \ell$  through  $|k - \ell|$  and is empty as soon as  $|k - \ell| > m$ . Thus it is a finite set, and moreover, by translation of the indices, for every  $k, \ell, m \in \mathbb{N}$  we have

(A.19) 
$$\int x^m \varphi_k(x) \varphi_\ell(x) \mu(\mathrm{d}x) = \mathbf{1}_{|k-\ell| \le m} \sum_{\gamma: (0,k-\ell) \to (m,0)} \prod_{e \in \gamma} \omega(e)_{\{(a_{n+\ell}),(b_{n+\ell})\}}$$

In particular (see (A.3)–(A.4)),

(A.20) 
$$\int x^m T_k(x) T_\ell(x) \mu_{\text{eq}}(dx) = \mathbf{1}_{|k-\ell| \le m} \sum_{\gamma: (0,k-\ell) \to (m,0)} \prod_{e \in \gamma} \omega(e)_{\{(a_{n+\ell}^*), (b_{n+\ell}^*)\}}.$$

Finally, by combining (A.19) and (A.20), we obtain

(A.21)  
$$\left| \int x^{m} \varphi_{k}(x) \varphi_{\ell}(x) \mu(\mathrm{d}x) - \int x^{m} T_{k}(x) T_{\ell}(x) \mu_{\mathrm{eq}}(\mathrm{d}x) \right|$$
$$\leq \sum_{\gamma:(0,k-\ell)\to(m,0)} \left| \prod_{e\in\gamma} \omega(e)_{\{(a_{n+\ell}),(b_{n+\ell})\}} - \prod_{e\in\gamma} \omega(e)_{\{(a_{n+\ell}^{*}),(b_{n+\ell}^{*})\}} \right|.$$

Together with the Nevai-class assumption for  $\mu$ , which states that  $a_n - a_n^* \to 0$  and  $b_n - b_n^* \to 0$  as  $n \to \infty$ , it follows that (A.15) and (A.16) hold true by taking  $k = \ell = N$ , or k = N and  $\ell = N - 1$ , in (A.21). This completes the proof of Lemma 1.  $\Box$ 

## A.2. Sketch of the proof of Theorem 2.1.

A.2.1. Reduction to probability reference measures. First, in the statement of Theorem 2.1, we can assume the reference measure  $\mu$  is a probability measure without loss of generality. This will simplify notation in the proof of Theorem 2.1.

Indeed, for any positive measure  $\mu$  on  $I^d$  with (multivariate) orthonormal polynomials  $\varphi_k$ and any  $\alpha > 0$ , the orthonormal polynomials associated with  $\alpha \mu$  are  $\varphi_k/\sqrt{\alpha}$ . Thus, if we momentarily denote by  $K_N(\mu; x, y)$  the Nth Christoffel–Darboux kernel associated with a measure  $\mu$ , we have  $K_N(\alpha\mu; x, y) = K_N(\mu; x, y)/\alpha$ . As a consequence, for every  $n \ge 1$ , the correlation measures

$$\det[K_N(\mu; x_i, x_\ell)]_{i,\ell=1}^n \prod_{i=1}^n \mu(\mathrm{d} x_i),$$

remain unchanged if we replace  $\mu$  by  $\alpha\mu$  for any  $\alpha > 0$ . Hence, multivariate OP Ensembles are invariant under  $\mu \mapsto \alpha\mu$ .

A.2.2. Soshnikov's key theorem. As stated previously, Theorem 2.1 has already been proven when d = 1 by [13], as a consequence of a generalized strong Szegő theorem they obtained. The difficulty in proving Theorem 2.1 when  $d \ge 2$  turns out to be of different nature than the one-dimensional setting. Indeed, the next result due to Soshnikov essentially

states that the cumulants of order three and more of the linear statistic  $\sum f(\mathbf{x}_i)$  decay to zero as  $N \to \infty$  as soon as its variance goes to infinity, and we will show the variance indeed diverges when  $d \ge 2$ . Thus, a CLT follows easily as soon as one can obtain asymptotic estimates on the variance. However, if obtaining such variance estimates is relatively easy when d = 1, the task becomes more involved in higher dimension.

More precisely, the general result [74], Theorem 1, has the following consequence.

THEOREM A.3 (Soshnikov). Let  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  form a multivariate OP Ensemble with respect to a given reference measure  $\mu$  on  $I^d$ . Consider a sequence  $(f_N)$  of uniformly bounded and measurable real-valued functions on  $I^d$  satisfying, as  $N \to \infty$ ,

(A.22) 
$$\mathbb{V}ar\left[\sum_{i=1}^{N} f_N(\mathbf{x}_i)\right] \longrightarrow \infty$$

and, for some  $\delta > 0$ ,

(A.23) 
$$\mathbb{E}\left[\sum_{i=1}^{N} |f_N(\mathbf{x}_i)|\right] = O\left(\mathbb{V}\mathrm{ar}\left[\sum_{i=1}^{N} f_N(\mathbf{x}_i)\right]^{\delta}\right).$$

Then we have

$$\frac{\sum_{i=1}^{N} f_N(\mathbf{x}_i) - \mathbb{E}[\sum_{i=1}^{N} f_N(\mathbf{x}_i)]}{\sqrt{\mathbb{V}\mathrm{ar}[\sum_{i=1}^{N} f_N(\mathbf{x}_i)]}} \xrightarrow[N \to \infty]{\mathrm{law}} \mathcal{N}(0, 1).$$

A.2.3. *Variance asymptotics*. In order to prove Theorem 2.1, it is enough to show the following asymptotics.

PROPOSITION 3. Assume  $\mu$  and  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  satisfy the hypothesis of Theorem 2.1. Then, for every  $f \in \mathcal{C}^1(I^d, \mathbb{R})$ , we have

(A.24) 
$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} f(\mathbf{x}_i)\right] = \sigma_f^2$$

Indeed, for any  $d \ge 2$  and any  $f \in \mathscr{C}^1(I^d, \mathbb{R})$ , Corollary 1 and Proposition 3 imply (A.22) and (A.23) with  $f_N = f$  and  $\delta = d/(d-1)$ . Thus, we can apply Theorem A.3 to obtain Theorem 2.1.

Proposition 3 is the main technical result of this work. Consider the *d*-fold product of the equilibrium measure (A.1), namely the probability measure on  $I^d$  given by

(A.25) 
$$\mu_{\text{eq}}^{\otimes d}(dx) = \omega_{\text{eq}}^{\otimes d}(x) dx, \qquad \omega_{\text{eq}}^{\otimes d}(x) = \prod_{j=1}^{d} \frac{1}{\pi \sqrt{1 - x_j^2}} \mathbf{1}_{I^d}(x).$$

In our proof of Proposition 3, we start by investigating the limit (A.24) when  $\mu = \mu_{eq}^{\otimes d}$ , since algebraic identities are available for this reference measure. Then we use comparison estimates to prove (A.24) in the general case.

A.3. A proof for the upper bound on the limiting variance. As stated in Proposition 1, one can bound the limiting variance  $\sigma_f^2$  by a Dirichlet energy. Besides providing some control on the amplitude of  $\sigma_f^2$ , we will need this inequality in the proof of Proposition 3. We now give a proof for this proposition.

PROOF OF PROPOSITION 1. Let  $\mu_{sc}(dx) = \pi^{-1}\sqrt{1-x^2}\mathbf{1}_I(x) dx$  be the semi-circle measure. The associated orthonormal polynomials are the so-called Chebyshev polynomials of the second kind

$$U_k(\cos\theta) = \sqrt{2} \frac{\sin((k+1)\theta)}{\sin\theta}.$$

For any  $1 \le j \le d$ , define the measure

$$\nu^{j}(\mathrm{d} x) = \mu_{\mathrm{eq}}(\mathrm{d} x_{1}) \cdots \mu_{\mathrm{eq}}(\mathrm{d} x_{j-1}) \mu_{\mathrm{sc}}(\mathrm{d} x_{j}) \mu_{\mathrm{eq}}(\mathrm{d} x_{j+1}) \cdots \mu_{\mathrm{eq}}(\mathrm{d} x_{d}),$$

so that the RHS of (2.9) becomes

$$\frac{1}{2}\sum_{j=1}^d \int_{I^d} (\partial_j f(x))^2 \nu^j (\mathrm{d} x).$$

For any  $\mathbf{k} = (k_1, \dots, k_d) \in \mathbb{N}^d$ , set  $T_{\mathbf{k}}(x) = T_{k_1}(x_1) \cdots T_{k_d}(x_d)$ , where  $T_k$  are the Chebyshev polynomials (A.2), and let

$$V_{k}^{j}(x) = T_{k_{1}}(x_{1}) \cdots T_{k_{j-1}}(x_{j-1}) U_{k_{j}}(x_{j}) T_{k_{j+1}}(x_{j+1}) \cdots T_{k_{d}}(x_{d}).$$

Thus,  $(T_k)_{k \in \mathbb{N}^d}$  and  $(V_k^j)_{k \in \mathbb{N}^d}$ , respectively, form an orthonormal Hilbert basis of  $L^2(\mu_{eq}^{\otimes d})$ and  $L^2(\nu^j)$ . Let  $f \in \mathscr{C}^1(I^d, \mathbb{R})$ , so that  $f = \sum_{k \in \mathbb{N}^d} \hat{f}(k) T_k$  where  $\hat{f}(k)$  is as in (2.8). Using the identity  $T'_k = kU_{k-1}$ , it comes

$$\partial_j f(x) = \sum_{\boldsymbol{k} \in \mathbb{N}^d} k_j \hat{f}(\boldsymbol{k}) V_{\boldsymbol{k}}^j(x)$$

Then Parseval's identity in  $L^2(v^j)$  yields

$$\int_{I^d} (\partial_j f(x))^2 v^j (\mathrm{d}x) = \sum_{\boldsymbol{k} \in \mathbb{N}^d} k_j^2 \hat{f}(\boldsymbol{k})^2.$$

Summing over  $1 \le j \le d$ , the RHS of (2.9) equals

(A.26) 
$$\frac{1}{2} \sum_{j=1}^{d} \int_{I^d} (\partial_j f(x))^2 v^j (dx) = \frac{1}{2} \sum_{k \in \mathbb{N}^d} (k_1^2 + \dots + k_d^2) \hat{f}(k)^2,$$

from which Proposition 1 easily follows.  $\Box$ 

**A.4.** Assumptions of Theorem 2.2 and outline of the proof. We now discuss the assumptions and proof of Theorem 2.2.

Assume the reference measure  $\mu$  is a product of d measures on I, and also that  $\mu$  has a density  $\omega$ . Then Corollary 1 suggests that, as  $N \to \infty$ ,

(A.27) 
$$\frac{N}{K_N(x,x)} \approx \frac{\omega(x)}{\omega_{\text{eq}}^{\otimes d}(x)}.$$

This heuristic would yield for the variance of the estimator (2.11),

(A.28) 
$$\mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} \frac{f(\mathbf{x}_{i})}{K_{N}(\mathbf{x}_{i},\mathbf{x}_{i})}\right] \approx \frac{1}{N^{2}} \mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} f(\mathbf{x}_{j}) \frac{\omega(\mathbf{x}_{j})}{\omega_{\operatorname{eq}}^{\otimes d}(\mathbf{x}_{j})}\right] \approx \frac{\Omega_{f,\omega}^{2}}{N^{1+1/d}},$$

where for the last approximation we used Proposition 3 with test function  $f \omega / \omega_{eq}^{\otimes d}$ , recalling  $\Omega_{f,\omega}$  was defined in (2.17). This would essentially yield the CLT in Theorem 2.2 by applying Theorem A.3 to  $f_N(x) = Nf(x)/K_N(x, x)$ . To make the approximation (A.28) rigorous, we will need extra regularity assumptions on  $\mu$ .

First, regarding the approximation (A.27), we have the following result.

THEOREM A.4 (Totik). Assume  $\mu(dx) = \omega(x) dx$  with  $\omega(x) = \omega_1(x_1) \cdots \omega_d(x_d)$ , and that  $\omega_j$  is continuous and positive on *I*. Then, for every  $\varepsilon > 0$ , we have

(A.29) 
$$\frac{N}{K_N(x,x)} \xrightarrow[N \to \infty]{} \frac{\omega(x)}{\omega_{eq}(x)}$$

uniformly for  $x \in I_{\varepsilon}^{d}$ .

For a proof of Theorem A.4 when d = 1, see [72], Section 3.11, and references therein. The case  $d \ge 2$  follows by the same arguments as in the proof of Corollary 1.

REMARK 7. It is because of Theorem A.4 that we restrict  $\mathscr{C}^1(I^d, \mathbb{R})$  to the class  $\mathscr{C}$  defined in (2.13) in the assumptions of Theorem 2.2. Unfortunately, there are examples of reference measures  $\mu$  on I such that the convergence (A.29) is not uniform on the whole of I. However, in order to extend  $\mathscr{C}$  to  $\mathscr{C}^1(I^d, \mathbb{R})$  in the statement of Theorem 2.2, it would be enough to have  $\sup_{x \in I_{\varepsilon_N}^d} |N/K_N(x, x) - \omega(x)/\omega_{eq}^{\otimes d}(x)| \to 0$  for some sequence  $\varepsilon_N$  going to zero as  $N \to \infty$ , but we were not able to locate such a result in the literature.

Next, the first approximation in (A.28) requires a control on the rate of change of  $N/K_N(x, x)$ . To this end, we introduce an extra assumption on the reference measure  $\mu$ . More precisely, let us denote

(A.30) 
$$\mathscr{D}_{N}(x, y) = \frac{N/K_{N}(x, x) - N/K_{N}(y, y)}{\|x - y\|}$$

and further consider the sequence of measures on  $I^d \times I^d$ 

(A.31) 
$$Q_N(dx, dy) = \frac{1}{N^{1-1/d}} \|x - y\|^2 K_N(x, y)^2 \mu(dx) \mu(dy).$$

Our extra assumption on  $\mu$  is then the following.

ASSUMPTION 1. The measure  $\mu$  satisfies

(A.32) 
$$\lim_{C \to \infty} \limsup_{\delta \to 0} \limsup_{N \to \infty} \iint_{I_{\varepsilon}^{d} \times I_{\varepsilon}^{d}, ||x-y|| \le \delta} \mathbf{1}_{|\mathscr{D}_{N}(x,y)| > C} \mathscr{D}_{N}(x,y)^{2} \times Q_{N}(dx, dy) = 0.$$

In plain words, this means the squared rate of change  $\mathscr{D}_N(x, y)^2$  is uniformly integrable with respect to the measures  $Q_N$ , at least on the restricted domain where ||x - y|| is small enough and where x and y are not allowed to reach the boundary of  $I^d$ .

REMARK 8. When d = 1, Lemma 1 states that if  $\mu$  is Nevai-class then  $Q_N$  converges weakly as  $N \to \infty$  towards

$$L(dx, dy) = \frac{1}{2\pi^2} \frac{1 - xy}{\sqrt{1 - x^2}\sqrt{1 - y^2}} \mathbf{1}_{I \times I}(x, y) dx dy.$$

Because the density of L is smooth within  $I_{\varepsilon} \times I_{\varepsilon}$  for every  $\varepsilon > 0$ , one may at least heuristically understand that (A.32) reduces to the uniform integrability of  $\mathscr{D}_N(x, y)^2$  with respect to the Lebesgue measure instead. In higher dimension, a similar guess can be made, but we do not pursue this reasoning here.

We now discuss sufficient conditions for (A.32) to hold true.

REMARK 9. Since, for any  $\kappa > 0$ , we have

$$\mathbf{1}_{|\mathscr{D}_N(x,y)|>C} \mathscr{D}_N(x,y)^2 \le \frac{1}{C^{\kappa}} |\mathscr{D}_N(x,y)|^{2+\kappa},$$

we see that condition (A.32) holds true as soon as, for every  $\varepsilon > 0$ , there exists  $\kappa, \delta > 0$  satisfying

$$\limsup_{N\to\infty}\iint_{I_{\varepsilon}^{d}\times I_{\varepsilon}^{d},\|x-y\|\leq\delta}|\mathscr{D}_{N}(x,y)|^{2+\kappa}Q_{N}(dx, dy)<\infty.$$

Namely, condition (A.32) is satisfied if the  $L^{2+\kappa}(Q_N)$  norm of the rate of change of  $N/K_N(x, x)$  is bounded, at least on the restricted domain where ||x - y|| is small enough and x, y away from the boundary of  $I^d$ .

The following assumption, which appears in Theorems 2.2 to 2.4, is much stronger than Assumption 1, but it is easier to check in practice.

ASSUMPTION 2. The measure  $\mu$  satisfies:

(a)  $\mu(dx) = \omega(x) dx$  with  $\omega$  positive and continuous on  $(-1, 1)^d$ .

(b) For every  $\varepsilon > 0$ , the sequence

$$\frac{1}{N}\sup_{x\in I_{\varepsilon}^{d}}\left\|\nabla K_{N}(x,x)\right\|$$

is bounded.

Indeed, thanks to the rough upper bound

$$\mathscr{D}_N(x, y) \Big| \le \sup_{x \in I_{\varepsilon}^d} \| \nabla (N/K_N(x, x)) \|, \quad x, y \in \mathbf{1}_{I_{\varepsilon}^d},$$

we see that Assumption 1 holds true as soon as for every  $\varepsilon > 0$ ,  $\sup_{x \in I_{\varepsilon}^{d}} \|\nabla(N/K_{N}(x, x))\|$  is bounded. Under Assumption 2(a), the latter follows from Assumption 2(b). Indeed, Theorem A.4 and Assumption 2(a) together yield that for every  $\varepsilon > 0$ , there exists c > 0 independent of N such that  $\frac{1}{N}K_{N}(x, x) > c$  for every  $x \in I_{\varepsilon}^{d}$ .

We conclude this section by proving that Jacobi measures (2.18) satisfy Assumption 2, which proves our Proposition 2. We start with a general lemma.

LEMMA 2. Assume the measures  $\mu_1, \ldots, \mu_d$  on I satisfy Assumption 2. Then the measure  $\mu_1 \otimes \cdots \otimes \mu_d$  on  $I^d$  satisfies Assumption 2.

PROOF. We decompose the set  $\Gamma_N = \{b(0), \ldots, b(N-1)\} \subset \mathbb{N}^d$  in a convenient way. To do so, set  $\sigma_j(\mathbf{k}) = (k_1, \ldots, k_{j-1}, k_{j+1}, \ldots, k_d)$  and say that  $\mathbf{k} \sim \boldsymbol{\ell}$  if and only if  $\sigma_j(\mathbf{k}) = \sigma_j(\boldsymbol{\ell})$ , that is, they have same coordinates except maybe the *j*th one. We denote by  $[\mathbf{k}]$  the equivalence class under this relation. Set  $N_j([\mathbf{k}]) = \max\{\ell_j : \boldsymbol{\ell} \in [\mathbf{k}] \cap \Gamma_N\}$ . Using the notation introduced in (A.7) and (A.8), it comes

(A.33)  

$$\partial_{j} K_{N}(x, x) = 2 \sum_{k=b(0)}^{b(N-1)} \varphi_{k_{j}}^{(j)}(x_{j}) \frac{d}{dx_{j}} \varphi_{k_{j}}^{(j)}(x_{j}) \prod_{\alpha \neq j} \varphi_{k_{\alpha}}^{(\alpha)}(x_{\alpha})^{2} \\
= 2 \sum_{[k] \in \Gamma_{N}/\sim \alpha \neq j} \prod_{\alpha \neq j} \varphi_{k_{\alpha}}^{(\alpha)}(x_{\alpha})^{2} \sum_{k_{j}=0}^{N_{j}([k])} \varphi_{k_{j}}^{(j)}(x_{j}) \frac{d}{dx_{j}} \varphi_{k_{j}}^{(j)}(x_{j}) \\
= \sum_{[k] \in \Gamma_{N}/\sim \alpha \neq j} \prod_{\alpha \neq j} \varphi_{k_{\alpha}}^{(\alpha)}(x_{\alpha})^{2} \frac{d}{dx_{j}} [K_{N_{j}([k])+1}^{(j)}(x_{j}, x_{j})].$$

Let now  $\varepsilon > 0$ . Since  $\mu_i$  satisfies Assumption 2, there exists C > 0 such that for all  $x \in I_{\varepsilon}$ and  $n \in \mathbb{N}$ ,

$$\left|\frac{\mathrm{d}}{\mathrm{d}x}\left[K_n^{(j)}(x,x)\right]\right| \le Cn.$$

Let  $M = \lfloor N^{1/d} \rfloor$ , so that  $\Gamma_N \subset \mathcal{C}_{M+1}$ , see (2.6). Thus,  $N_i([k]) \leq M$  for all  $k \in \Gamma_N$ . By (A.33),

$$\begin{aligned} \left|\partial_{j}K_{N}(x,x)\right| &\leq C(M+1)\sum_{[k]\in\Gamma_{N}/\sim\alpha\neq j}\prod_{\alpha\neq j}\varphi_{k_{\alpha}}^{(\alpha)}(x_{\alpha})^{2} \\ &\leq C(M+1)\sum_{[k]\in\mathcal{C}_{M+1}}\prod_{\alpha\neq j}\varphi_{k_{\alpha}}^{(\alpha)}(x_{\alpha})^{2} \\ &= C(M+1)\prod_{\alpha\neq j}K_{M+1}^{(\alpha)}(x_{\alpha},x_{\alpha}). \end{aligned}$$

Hence

$$\frac{1}{N} \left| \partial_j K_N(x, x) \right| \le C \frac{M+1}{M} \prod_{\alpha \neq j} \frac{1}{M} K_{M+1}^{(\alpha)}(x_\alpha, x_\alpha),$$

and the lemma follows with Theorem A.4.  $\Box$ 

LEMMA 3. Let  $\alpha$ ,  $\beta > -1$ , then the measure

$$(1-x)^{\alpha}(1+x)^{\beta}\mathbf{1}_{I}(x)\,\mathrm{d}x$$

satisfies Assumption 2.

**PROOF.** Let  $\varepsilon > 0$  be fixed. For convenience, Section A.2.1 allows us to work with the probability measure

$$\mu^{(\alpha,\beta)}(\mathrm{d}x) = \omega^{(\alpha,\beta)}(x)\,\mathrm{d}x, \qquad \omega^{(\alpha,\beta)}(x) = \frac{1}{c_{\alpha,\beta}}(1-x)^{\alpha}(1+x)^{\beta},$$

where the normalization constant reads

$$c_{\alpha,\beta} = 2^{\alpha+\beta+1} \frac{\Gamma(\alpha+1)\Gamma(\beta+1)}{\Gamma(\alpha+\beta+1)}$$

and  $\Gamma$  is the Euler Gamma function. Denote by  $(\varphi_n^{(\alpha,\beta)})_{n\in\mathbb{N}}$  the associated orthonormal polynomials. They satisfy

$$\varphi_n^{(\alpha,\beta)}(x) = \frac{P_n^{(\alpha,\beta)}(x)}{\sqrt{h_n^{(\alpha,\beta)}}},$$

where the  $P_n^{(\alpha,\beta)}$ 's are the Jacobi polynomials (we refer to [76] for definitions and properties) and

$$\begin{split} h_n^{(\alpha,\beta)} &= \|P_n^{(\alpha,\beta)}\|_{L^2(\mu^{(\alpha,\beta)})}^2 \\ &= \frac{1}{n!(\alpha+\beta+2n+1)} \frac{\Gamma(\alpha+\beta+1)\Gamma(\alpha+n+1)\Gamma(\beta+n+1)}{\Gamma(\alpha+1)\Gamma(\beta+1)\Gamma(\alpha+\beta+n+1)}, \end{split}$$

and moreover

(A.34)  
$$(\varphi_n^{(\alpha,\beta)})' = \frac{u_{\alpha,\beta}}{2} \sqrt{n(n+\alpha+\beta+1)} \varphi_{n-1}^{(\alpha+1,\beta+1)},$$
$$u_{\alpha,\beta} = \sqrt{\frac{(\alpha+\beta+1)(\alpha+\beta+2)}{(\alpha+1)(\beta+1)}}.$$

This yields

(A.35)  
$$\frac{\mathrm{d}}{\mathrm{d}x}K_N(x,x) = 2\sum_{k=1}^{N-1} \varphi_k^{(\alpha,\beta)}(x) (\varphi_k^{(\alpha,\beta)})'(x) \\ = u_{\alpha,\beta} \sum_{k=1}^{N-1} \sqrt{k(k+\alpha+\beta+1)} \varphi_k^{(\alpha,\beta)}(x) \varphi_{k-1}^{(\alpha+1,\beta+1)}(x).$$

By [45], we have as  $k \to \infty$ , uniformly in  $x = \cos \theta \in I_{\varepsilon}$ ,

(A.36)  
$$\varphi_{k}^{(\alpha,\beta)}(\cos\theta) = \sqrt{\frac{2}{\omega^{(\alpha,\beta)}(x)\pi\sqrt{1-x^{2}}}} \cos\left(\left(k + \frac{1}{2}(\alpha+\beta+1)\right)\theta - \frac{\pi}{2}\left(\alpha + \frac{1}{2}\right)\right) + O(1/k).$$

As a consequence, we obtain in the same asymptotic regime,

(A.37) 
$$\varphi_{k-1}^{(\alpha+1,\beta+1)}(\cos\theta) = \sqrt{\frac{2}{\omega^{(\alpha,\beta)}(x)\pi\sqrt{1-x^2}}}\sin\left(\left(k+\frac{1}{2}(\alpha+\beta+1)\right)\theta - \frac{\pi}{2}\left(\alpha+\frac{1}{2}\right)\right) + O(1/k).$$

Now (A.36) implies that the  $P_k^{(\alpha,\beta)}(x)$ 's are bounded uniformly for  $x \in I$  and  $k \in \mathbb{N}$ . Using moreover that  $2\sin(u)\cos(u) = \sin(2u)$  and combining (A.35)–(A.37), we obtain for some  $C_1, C_2 > 0$  that

$$\sup_{x\in I_{\varepsilon}}\left|\frac{\mathrm{d}}{\mathrm{d}x}K_{N}(x,x)\right| \leq C_{1}\sup_{x\in I_{\varepsilon}}\left|\sum_{k=1}^{N-1}k\sin\left((2k+\alpha+\beta+1)\theta-\pi\left(\alpha+\frac{1}{2}\right)\right)\right|+C_{2},$$

where we recall the relation  $x = \cos \theta$ . Next, we write

$$\left|\sum_{k=1}^{N-1} k \sin\left((2k+\alpha+\beta+1)\theta - \pi\left(\alpha+\frac{1}{2}\right)\right)\right| \le \left|\sum_{k=1}^{N-1} k e^{i\left((2k+\alpha+\beta+1)\theta - \pi\left(\alpha+\frac{1}{2}\right)\right)}\right|^{1/2}$$

and then

(A.38)  

$$\sum_{k=1}^{N-1} k e^{i((2k+\alpha+\beta+1)\theta-\pi(\alpha+\frac{1}{2}))}$$

$$= e^{i((\alpha+\beta+1)\theta-\pi(\alpha+\frac{1}{2}))} \sum_{k=1}^{N-1} k e^{2ik\theta}$$

$$= \frac{1}{2i} e^{i((\alpha+\beta+1)\theta-\pi(\alpha+\frac{1}{2}))} \frac{\mathrm{d}}{\mathrm{d}\theta} \sum_{k=0}^{N-1} e^{2ik\theta}$$

$$= \frac{1}{2i} e^{i((\alpha+\beta+1)\theta-\pi(\alpha+\frac{1}{2}))} \frac{\mathrm{d}}{\mathrm{d}\theta} \left(\frac{e^{i(N-1)\theta}\sin(N\theta)}{\sin(\theta)}\right).$$

Since the absolute value of the right-hand side of (A.38) is bounded by  $CN/\sin^2(\theta)$  for some C > 0 independent on N and  $\theta$ , the lemma follows.  $\Box$ 

Lemmas 2 and 3 combined yield Proposition 2.

## A.5. Proof of Theorem 2.4.

PROOF OF THEOREM 2.4. We proceed in two steps. The first step is to prove a bivariate CLT for the vector  $(\mathscr{I}_N(f), \mathscr{I}_N(1))$ , and the second step is to apply the delta method to obtain a central limit theorem for the ratio.

Let  $t, v \in \mathbb{R}$ . We apply Theorem 2.3 to tf + vh, where h is any function in  $\mathscr{C}$  that is equal to 1 on  $I_{\varepsilon}^{d}$ . This leads to

$$\sqrt{N^{1+1/d}} \left( t \mathscr{I}_N(f) + v \mathscr{I}_N(1) - t \int f \omega_u \, \mathrm{d}x - v Z \right) \to \mathcal{N}(0, Z^2 \Omega_{tf+vh,\omega}^2),$$

where

$$\Omega_{tf+vh,\omega}^2 = t^2 \Omega_{f,\omega}^2 + v^2 \Omega_{1,\omega}^2 + 2tvc_{f,\omega}.$$

By the Crámer–Wold theorem [41], Corollary 5.5, we obtain a bivariate CLT

$$\sqrt{N^{1+1/d}} \left[ \begin{pmatrix} \mathscr{I}_N(f) \\ \mathscr{I}_N(1) \end{pmatrix} - \begin{pmatrix} \int f \omega_u \, \mathrm{d}x \\ Z \end{pmatrix} \right] \to \mathcal{N} \left( 0, Z^2 \begin{pmatrix} \Omega_{f,\omega}^2 & c_{f,\omega} \\ c_{f,\omega} & \Omega_{1,\omega}^2 \end{pmatrix} \right).$$

Now we apply the delta method [70], Section 7.1.3, to obtain

$$\begin{split} \sqrt{N^{1+1/d}} & \left[ \frac{\mathscr{I}_N(f)}{\mathscr{I}_N(1)} - \int f \, \mathrm{d}\mu \right] \\ \to \mathcal{N} & \left( 0, Z^2 \begin{pmatrix} \frac{1}{Z} & -\frac{\int f \omega_u \, \mathrm{d}x}{Z^2} \end{pmatrix} \begin{pmatrix} \Omega_{f,\omega}^2 & c_{f,\omega} \\ c_{f,\omega} & \Omega_{1,\omega}^2 \end{pmatrix} \begin{pmatrix} \frac{1}{Z} \\ -\frac{\int f \omega_u \, \mathrm{d}x}{Z^2} \end{pmatrix} \right) \end{split}$$

which is precisely (2.22).  $\Box$ 

## APPENDIX B: CLT FOR MULTIVARIATE OP ENSEMBLES: PROOF OF THEOREM 2.1

In this section, we prove Proposition 3. As explained in Section A.2, Theorem 2.1 follows from this proposition.

## **B.1.** A useful representation of the covariance.

LEMMA 4. Let  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  be random variables drawn from a multivariate OP Ensemble with reference measure  $\mu$ . For any multivariate polynomials P, Q, we have

$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} P(\mathbf{x}_{i}), \sum_{i=1}^{N} Q(\mathbf{x}_{i})\right] = \sum_{n=0}^{N-1} \sum_{m=N}^{\infty} \langle P\varphi_{n}, \varphi_{m} \rangle \langle Q\varphi_{n}, \varphi_{m} \rangle,$$

where  $\langle \cdot, \cdot \rangle$  refers to the scalar product of  $L^2(\mu)$ .

PROOF. We start from the standard formula

(B.1)  

$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} P(\mathbf{x}_{i}), \sum_{j=1}^{N} Q(\mathbf{x}_{i})\right]$$

$$= \int P(x)Q(x)K_{N}(x, x)\mu(dx)$$

$$- \iint P(x)Q(y)K_{N}(x, y)^{2}\mu(dx)\mu(dy)$$

which follows from (2.1)–(2.3) with n = 1, 2 and that  $K_N(x, y)$  is symmetric. On the one hand, it follows from the definition of  $K_N$  that

(B.2) 
$$\iint P(x)Q(y)K_N(x,y)^2\mu(\mathrm{d}x)\mu(\mathrm{d}y) = \sum_{n=0}^{N-1}\sum_{m=0}^{N-1} \langle P\varphi_n,\varphi_m\rangle\langle Q\varphi_n,\varphi_m\rangle.$$

On the other hand, by using the decomposition (where the sum is finite since P is polynomial)

$$P\varphi_n = \sum_{m=0}^{\infty} \langle P\varphi_n, \varphi_m \rangle \varphi_m$$

together with the identity

$$\int P(x)Q(x)K_N(x,x)\mu(\mathrm{d}x) = \sum_{n=0}^{N-1} \langle P\varphi_n, Q\varphi_n \rangle,$$

we obtain

(B.3) 
$$\int P(x)Q(x)K_N(x,x)\mu(dx) = \sum_{n=0}^{N-1} \sum_{m=0}^{\infty} \langle P\varphi_n,\varphi_m \rangle \langle Q\varphi_n,\varphi_m \rangle.$$

Lemma 4 then follows by combining (B.1), (B.2) and (B.3).  $\Box$ 

**B.2.** Covariance asymptotics: The Chebyshev case. We first investigate the case of the product measure  $\mu_{eq}^{\otimes d}$ , where  $\mu_{eq}$  defined in (A.1) is the equilibrium measure of *I*. Recalling the definition (A.2), the multivariate Chebyshev polynomials

(B.4) 
$$T_{k}(x_{1},...,x_{d}) = T_{k_{1}}(x_{1})\cdots T_{k_{d}}(x_{d}), \quad k = (k_{1},...,k_{d}) \in \mathbb{N}^{d},$$

satisfy the orthonormality conditions

$$\int T_{\boldsymbol{k}}(x)T_{\boldsymbol{\ell}}(x)\mu_{\mathrm{eq}}^{\otimes d}(\mathrm{d} x) = \delta_{\boldsymbol{k}\boldsymbol{\ell}}, \quad \boldsymbol{k}, \boldsymbol{\ell} \in \mathbb{N}^{d}$$

We shall see that the family  $(T_k)_{k \in \mathbb{N}^d}$  diagonalizes the covariance structure associated with our point process.

PROPOSITION 4. Let  $\mathbf{x}_1^*, \ldots, \mathbf{x}_N^*$  be drawn according to the multivariate OP Ensemble associated with  $\mu_{eq}^{\otimes d}$ . Then, given any multi-indices  $\mathbf{k} = (k_1, \ldots, k_d) \in \mathbb{N}^d$  and  $\boldsymbol{\ell} = (\ell_1, \ldots, \ell_d) \in \mathbb{N}^d$ , we have

$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{C}ov \left[ \sum_{i=1}^{N} T_{\boldsymbol{k}}(\mathbf{x}_{i}^{*}), \sum_{i=1}^{N} T_{\boldsymbol{\ell}}(\mathbf{x}_{i}^{*}) \right] = \begin{cases} \frac{1}{2}(k_{1} + \dots + k_{d}) & \text{if } \boldsymbol{k} = \boldsymbol{\ell} \\ 0 & \text{if } \boldsymbol{k} \neq \boldsymbol{\ell} \end{cases}$$

As a warm-up, let us first prove the proposition when d = 1.

PROOF OF PROPOSITION 4 WHEN d = 1. Throughout this proof,  $\langle \cdot, \cdot \rangle$  denotes the inner product in  $L^2(\mu_{eq}^{\otimes d})$ . For every  $k, \ell \in \mathbb{N}$ , Lemma 4 provides

(B.5) 
$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} T_{k}(\mathbf{x}_{i}^{*}), \sum_{i=1}^{N} T_{\ell}(\mathbf{x}_{i}^{*})\right] = \sum_{n=0}^{N-1} \sum_{m=N}^{\infty} \langle T_{k}T_{n}, T_{m} \rangle \langle T_{\ell}T_{n}, T_{m} \rangle$$

First, notice that if k or  $\ell$  is zero, then the right-hand side of (B.5) vanishes because  $\langle T_n, T_m \rangle = \delta_{nm}$ , and hence we can assume both k,  $\ell$  are nonzero. Next, (A.2) yields the multiplication formula

(B.6) 
$$T_k T_n = \frac{1}{\sqrt{2}} T_{n+k} \mathbf{1}_{kn\neq 0} + \left(\frac{1}{\sqrt{2}}\right)^{\mathbf{1}_{nk\neq 0} \mathbf{1}_{n\neq k}} T_{|n-k|}, \quad k, n \in \mathbb{N}.$$

Combined with the orthonormality relations, this yields for any  $n, m \in \mathbb{N}$  and k > 0

(B.7) 
$$\langle T_k T_n, T_m \rangle = \frac{1}{\sqrt{2}} \mathbf{1}_{n+k=m} \mathbf{1}_{n\neq 0} + \left(\frac{1}{\sqrt{2}}\right)^{\mathbf{1}_{n\neq 0} \mathbf{1}_{n\neq k}} \mathbf{1}_{|n-k|=m}.$$

Hence, if  $n, m \in \mathbb{N}$  moreover satisfy n < m and  $m > \max(k, \ell)$ , then we have

(B.8) 
$$\langle T_k T_n, T_m \rangle \langle T_\ell T_n, T_m \rangle = \frac{1}{2} \mathbf{1}_{n \neq 0} \mathbf{1}_{n+k=m} \mathbf{1}_{\ell+n=m}.$$

By plugging (B.8) into (B.5), we obtain for every  $N > \max(k, \ell)$ ,

$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} T_{k}(\mathbf{x}_{i}^{*}), \sum_{i=1}^{N} T_{\ell}(\mathbf{x}_{i}^{*})\right] = \frac{1}{2} \sum_{n=1}^{N-1} \sum_{m=N}^{\infty} \mathbf{1}_{k+n=m} \mathbf{1}_{\ell+n=m}$$
$$= \frac{1}{2} k \mathbf{1}_{k=\ell},$$

and the proposition follows when d = 1.  $\Box$ 

We now provide a proof for the higher-dimensional case. We also use the multiplication formula (B.6) in an essential way, although the setting is more involved. We recall that we introduced the bijection  $\mathfrak{b} : \mathbb{N} \to \mathbb{N}^d$  associated with the graded lexicographic order in Section 2.1.3.

PROOF OF PROPOSITION 4 WHEN  $d \ge 2$ . Fix multi-indices  $\mathbf{k} = (k_1, \dots, k_d) \in \mathbb{N}^d$  and  $\boldsymbol{\ell} = (\ell_1, \dots, \ell_d) \in \mathbb{N}^d$ , and also set

$$S = \{j : k_j \neq 0\}, \qquad S' = \{j : \ell_j \neq 0\}.$$

Thanks to Lemma 4, we can write

(B.9) 
$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} T_{k}(\mathbf{x}_{i}^{*}), \sum_{i=1}^{N} T_{\ell}(\mathbf{x}_{i}^{*})\right] = \sum_{(n,m)\in\mathbb{A}_{N}} \langle T_{k}T_{n}, T_{m}\rangle \langle T_{\ell}T_{n}, T_{m}\rangle,$$

where we introduced for convenience the set

(B.10) 
$$\mathbb{A}_N = \{ (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{N}^d \times \mathbb{N}^d : \boldsymbol{n} \le \mathfrak{b}(N-1), \boldsymbol{m} \ge \mathfrak{b}(N) \}.$$

Next, using (B.4), the orthonormality relations for the Chebyshev polynomials and (B.7), we obtain

(B.11) 
$$\langle T_{k}T_{n}, T_{m} \rangle = \langle T_{k_{1}}T_{n_{1}}, T_{m_{1}} \rangle_{L^{2}(\mu_{eq})} \cdots \langle T_{k_{d}}T_{n_{d}}, T_{m_{d}} \rangle_{L^{2}(\mu_{eq})}$$
$$= \left(\prod_{j \notin S} \mathbf{1}_{n_{j}=m_{j}}\right) \sum_{P \subset S} \left(\frac{1}{\sqrt{2}}\right)^{|P|+|\{j \in S \setminus P: n_{j} \neq 0, n_{j} \neq k_{j}\}|}$$
$$\times \left(\prod_{j \in P} \mathbf{1}_{n_{j}+k_{j}=m_{j}} \mathbf{1}_{n_{j}\neq 0}\right) \left(\prod_{j \in S \setminus P} \mathbf{1}_{|n_{j}-k_{j}|=m_{j}}\right),$$

where |A| stands for the cardinality of the set A.

First, notice that if  $S \neq S'$  then the right-hand side of (B.9) vanishes. Indeed, if  $S \neq S'$ , then there exists  $\alpha \in \{1, \ldots, d\}$  such that  $k_{\alpha} = 0$  and  $\ell_{\alpha} \neq 0$  (or the other way around, but the argument is symmetric). It then follows from (B.12) that  $\langle T_k T_n, T_m \rangle$  vanishes except if  $n_{\alpha} = m_{\alpha}$ , and moreover that  $\langle T_{\ell}T_n, T_m \rangle$  vanishes except if  $|n_{\alpha} \pm \ell_{\alpha}| = m_{\alpha}$ . Since  $\ell_{\alpha} \neq 0$ , it holds  $\langle T_k T_n, T_m \rangle \langle T_{\ell}T_n, T_m \rangle = 0$  for every  $(n, m) \in \mathbb{N}^d \times \mathbb{N}^d$ , and our claim follows. Moreover, because  $(n, m) \in \mathbb{A}_N$  yields the existence of  $\alpha \in \{1, \ldots, d\}$  such that  $n_{\alpha} < m_{\alpha}$ , one can see from (B.11) that  $\langle T_k T_n, T_m \rangle$  vanishes for every  $(n, m) \in \mathbb{A}_N$  if  $\mathbf{k} = (0, \ldots, 0)$ . We henceforth assume that  $S = S' \neq \emptyset$ , for the covariance not to be trivial.

By combining (B.9) with (B.12), we obtain

$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} T_{k}(\mathbf{x}_{i}^{*}), \sum_{i=1}^{N} T_{\ell}(\mathbf{x}_{i}^{*})\right] = \sum_{P, Q \subset S} \sum_{(n,m) \in \mathbb{A}_{N}[P,Q]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,Q](n)},$$

where we introduced the subsets

(B.13) 
$$\mathbb{A}_{N}[P,Q] = \left\{ (\boldsymbol{n},\boldsymbol{m}) \in \mathbb{A}_{N} \middle| \begin{array}{l} n_{j} + k_{j} = m_{j}, & n_{j} \neq 0, & \text{if } j \in P \\ n_{j} + \ell_{j} = m_{j}, & n_{j} \neq 0, & \text{if } j \in Q \\ |n_{j} - k_{j}| = m_{j}, & \text{if } j \in S \setminus P \\ |n_{j} - \ell_{j}| = m_{j}, & \text{if } j \in S \setminus Q \end{array} \right. \quad n_{j} = m_{j}, \quad \text{if } j \notin S \left. \right\}$$

and set for convenience

(B.14) 
$$\sigma[P, Q](\mathbf{n}) = |P| + |Q| + |\{j \in S \setminus P : n_j \neq 0, n_j \neq k_j\}| + |\{j \in S \setminus Q : n_j \neq 0, n_j \neq \ell_j\}|.$$

Notice from (B.13) if  $k_{\alpha} = \ell_{\alpha} \neq 0$  and  $\mathbb{A}_{N}[P, Q] \neq \emptyset$  then necessarily  $\alpha \in P \cap Q$  or  $\alpha \in (S \setminus P) \cap (S \setminus Q)$ . In particular, if  $\mathbf{k} = \boldsymbol{\ell}$  then  $\mathbb{A}_{N}[P, Q] = \emptyset$  unless P = Q. Thus,

(B.15)  

$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} T_{k}(\mathbf{x}_{i}^{*}), \sum_{i=1}^{N} T_{\ell}(\mathbf{x}_{i}^{*})\right] = \mathbf{1}_{k=\ell} \sum_{P \subset S} \sum_{(n,m) \in \mathbb{A}_{N}[P,P]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,P](n)} + \mathbf{1}_{k\neq\ell} \sum_{P,Q \subset S} \sum_{(n,m) \in \mathbb{A}_{N}[P,Q]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,Q](n)}$$

Our goal is now to show that for every  $P, Q \subset S$  the following holds true. As  $N \to \infty$ , if we assume  $k = \ell$ , then

(B.16) 
$$\sum_{(\boldsymbol{n},\boldsymbol{m})\in\mathbb{A}_{N}[P,P]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,P](\boldsymbol{n})} = \left(\frac{1}{2}\right)^{|S|} \left(\sum_{j\in P} k_{j}\right) N^{1-1/d} + o(N^{1-1/d}),$$

and, if instead  $k \neq \ell$ , then

(B.17) 
$$|\mathbb{A}_N[P,Q]| = o(N^{1-1/d}).$$

Since an easy rearrangement argument together with the definition of S yield

$$\sum_{P \subset S} \left( \sum_{j \in P} k_j \right) = \frac{1}{2} \sum_{P \subset S} \left( \sum_{j \in S} k_j \right)$$
$$= \left( \sum_{j \subset S} k_j \right) 2^{|S|-1} = \left( \sum_{j=1}^d k_j \right) 2^{|S|-1},$$

Proposition 4 would then follow from (B.15)–(B.17).

We now turn to the proof of (B.16) and (B.17).

*Truncated sets and consequences.* Given distinct  $\alpha_1, \ldots, \alpha_p \in \{1, \ldots, d\}$ , we introduce the truncated sets

(B.18) 
$$\mathbb{A}_{N}[P, Q; \alpha_{1}, \dots, \alpha_{p}]$$
$$= \mathbb{A}_{N}[P, Q] \cap \{n_{\alpha_{1}} \leq \max(k_{\alpha_{1}}, \ell_{\alpha_{1}})\} \cap \dots \cap \{n_{\alpha_{p}} \leq \max(k_{\alpha_{p}}, \ell_{\alpha_{p}})\}$$

By definition of b and  $\mathbb{A}_N$ , if  $N = M^d$ , then  $\mathbb{A}_{M^d} = \mathcal{C}_M \times (\mathbb{N}^d \setminus \mathcal{C}_M)$  where we recall

(B.19) 
$$\mathcal{C}_M = \{ \boldsymbol{n} \in \mathbb{N}^d : 0 \le n_1, \dots, n_d \le M - 1 \}.$$

Moreover, if for an arbitrary N we denote by  $M = \lfloor N^{1/d} \rfloor$  the integer satisfying  $M^d \leq N < (M + 1)^d$ , then  $\mathfrak{b}(N) \in \mathcal{C}_{M+1}$ , and thus, for any  $(n, m) \in \mathbb{A}_N$ , we have  $n \in \mathcal{C}_{M+1}$ . As a consequence, for every  $P, Q \subset S$ , we have the rough upper bound  $|\mathbb{A}_N[P, Q; \alpha_1, \dots, \alpha_p]| = \mathcal{O}(M^{d-p})$ . In particular,

(B.20) 
$$|\mathbb{A}_N[P,Q;\alpha_1,\ldots,\alpha_p]| = o(N^{1-1/d})$$
 for every  $p \ge 2$ .

In order to restrict ourselves to the easier setting where N is a power of d, we will use the following lemma. Its proof uses in a crucial way the graded lexicographic order we chose to equip  $\mathbb{N}^d$  with, and it is deferred to the end of the present proof.

LEMMA 5. Assume  $\mathbf{k} = \mathbf{\ell}$ . For every  $P \subset S$ ,  $\alpha \in S \setminus P$  and for every  $N > \max(k_1^d, \ldots, k_d^d)$ , we have:

- (a)  $|\mathbb{A}_N[P, P]| \le |\mathbb{A}_{N+1}[P, P]|,$
- (b)  $|\mathbb{A}_{N}[P, P; \alpha]| \le |\mathbb{A}_{N+1}[P, P; \alpha]|.$

*Proof of* (B.16). Assume  $\mathbf{k} = \mathbf{\ell}$ . As a consequence of Lemma 5 (a), if we set  $M = \lfloor N^{1/d} \rfloor$  then we have for every N large enough

$$\sum_{(n,m)\in\mathbb{A}_{M^{d}}[P,P]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,P](n)} \leq \sum_{(n,m)\in\mathbb{A}_{N}[P,P]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,P](n)}$$
$$\leq \sum_{(n,m)\in\mathbb{A}_{(M+1)^{d}}[P,P]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,P](n)}.$$

Thus, it is enough to prove that, as  $M \to \infty$ ,

(B.21) 
$$\sum_{(\boldsymbol{n},\boldsymbol{m})\in\mathbb{A}_{M^{d}}[P,P]} \left(\frac{1}{\sqrt{2}}\right)^{\sigma[P,P](\boldsymbol{n})} = \left(\frac{1}{2}\right)^{|S|} \left(\sum_{j\in P} k_{j}\right) M^{d-1} + o(M^{d-1}),$$

in order to establish (B.16). To do so, for any  $P \subset S$  and  $\alpha \in S \setminus P$ , we set

(B.22) 
$$\mathbb{A}_{M^d}^*[P] = \mathbb{A}_{M^d}[P, P] \cap \{n_j > k_j \text{ for all } j \in S \setminus P\},$$

(B.23) 
$$\mathbb{A}^*_{M^d}[P;\alpha] = \mathbb{A}_{M^d}[P,P;\alpha] \cap \{n_j > k_j \text{ for all } j \in S \setminus (P \cup \{\alpha\})\},$$

and use the following lemma; its proof is deferred to the end of the actual proof.

LEMMA 6. Assume  $\mathbf{k} = \mathbf{\ell}$ . For every  $P \subset S$  and  $\alpha \in S \setminus P$ , we have as  $M \to \infty$ ,

(B.24) 
$$|\mathbb{A}^*_{M^d}[P]| = \left(\sum_{j \in P} k_j\right) M^{d-1} + o(M^{d-1}),$$

(B.25)  $\left|\mathbb{A}_{M^d}^*[P;\alpha]\right| = o(M^{d-1}).$ 

Next, as a consequence of the rough upper bound (B.20) and (B.25), we can write

$$|\mathbb{A}_{M^{d}}[P, P]| = |\mathbb{A}_{M^{d}}[P, P] \cap \{n_{\alpha} > k_{\alpha} \text{ for all } \alpha \in S \setminus P\}|$$
  
+  $|\mathbb{A}_{M^{d}}[P, P] \cap \{n_{\alpha} \le k_{\alpha} \text{ for at least one } \alpha \in S \setminus P\}|$   
(B.26)  
$$= |\mathbb{A}_{M^{d}}^{*}[P]| + \sum_{\alpha \in S \setminus P} |\mathbb{A}_{M^{d}}^{*}[P; \alpha]| + o(M^{d-1})$$
  
$$= |\mathbb{A}_{M^{d}}^{*}[P]| + o(M^{d-1}).$$

Since for any  $(n, m) \in \mathbb{A}^*_{M^d}[P]$  we have  $\sigma[P, P](n) = 2|S|$ , see (B.14) and (B.22), the estimate (B.21) follows from (B.26) and (B.24), and the proof of (B.16) is therefore complete.

*Proof of* (B.17). Assume now that  $k \neq \ell$ . Since k and  $\ell$  have the same zero components, it follows that neither  $k_{\alpha}$  nor  $\ell_{\alpha}$  is zero. Thus, (B.13) yields that if  $k_{\alpha} \neq \ell_{\alpha}$  and  $\mathbb{A}_{N}[P, Q] \neq \emptyset$ , then either  $\alpha \in P \cap (S \setminus Q)$  or  $\alpha \in Q \cap (S \setminus P)$ , and moreover, for any  $(n, m) \in \mathbb{A}_{N}[P, Q]$ , we have

$$2n_{\alpha} = |k_{\alpha} - \ell_{\alpha}|, \qquad 2m_{\alpha} = k_{\alpha} + \ell_{\alpha}.$$

In particular,  $\mathbb{A}_{M^d}[P, Q] = \mathbb{A}_{M^d}[P, Q; \alpha]$ . Thus, by virtue of the rough upper bound (B.20), we can assume in the proof of (B.17) that  $\boldsymbol{k}$  and  $\boldsymbol{\ell}$  differ by exactly one coordinate, namely there exists  $\alpha \in \{1, \ldots, d\}$  such that  $k_{\alpha} \neq \ell_{\alpha}$  and  $k_j = \ell_j$  for every  $j \neq \alpha$ . In this setting,  $\mathbb{A}_N[P, Q] \neq \emptyset$  then yields  $P \setminus \{\alpha\} = Q \setminus \{\alpha\}$  and, if  $(\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N[P, Q]$ , then  $(n_{\alpha}, m_{\alpha})$  satisfies the equations

$$\begin{cases} n_{\alpha} + k_{\alpha} = m_{\alpha}, & \ell_{\alpha} - n_{\alpha} = m_{\alpha}, & n_{\alpha} \neq 0, & \text{if } \alpha \in P, \\ n_{\alpha} + \ell_{\alpha} = m_{\alpha}, & k_{\alpha} - n_{\alpha} = m_{\alpha}, & n_{\alpha} \neq 0, & \text{if } \alpha \in Q. \end{cases}$$

By weakening these constraints to

$$\begin{cases} |\ell_{\alpha} - n_{\alpha}| = m_{\alpha}, & n_{\alpha} \le \ell_{\alpha} & \text{if } \alpha \in P, \\ |k_{\alpha} - n_{\alpha}| = m_{\alpha}, & n_{\alpha} \le k_{\alpha} & \text{if } \alpha \in Q, \end{cases}$$

we obtain the upper bound

(B.27) 
$$|\mathbb{A}_{N}[P,Q]| \leq \begin{cases} |\mathbb{A}_{N}^{(\ell,\ell)}[Q,Q;\alpha]| & \text{if } \alpha \in P, \\ |\mathbb{A}_{N}^{(k,k)}[P,P;\alpha]| & \text{if } \alpha \in Q, \end{cases}$$

where  $\mathbb{A}_N^{(k,\ell)}[P, Q; \alpha]$  is defined as in (B.13), (B.18) but we emphasized the multi-indices k,  $\ell$  which are involved. By setting  $M = \lfloor N^{1/d} \rfloor + 1$ , we obtain from Lemma 5(b), the rough upper bound (B.20) and (B.23), (B.25) that, as  $N \to \infty$ ,

(B.28)  

$$|\mathbb{A}_{N}^{(\boldsymbol{k},\boldsymbol{k})}[P, P; \alpha]|$$

$$\leq |\mathbb{A}_{M^{d}}^{(\boldsymbol{k},\boldsymbol{k})}[P, P; \alpha]|$$

$$= |\mathbb{A}_{M^{d}}^{(\boldsymbol{k},\boldsymbol{k})}[P, P; \alpha] \cap \{n_{j} > k_{j} \text{ for all } j \in S \setminus (P \cup \{\alpha\})\}| + o(N^{1-1/d})$$

$$= o(N^{1-1/d}),$$

and similarly

(B.29) 
$$\left|\mathbb{A}_{N}^{(\boldsymbol{\ell},\boldsymbol{\ell})}[Q,Q;\alpha]\right| = o(N^{1-1/d}).$$

.....

By combining (B.27)–(B.29), we have finally proved (B.17) and the proof of Proposition 4 is thus complete, up to the proof of Lemmas 5 and 6.  $\Box$ 

We now provide proofs for the remaining lemmas.

PROOF OF LEMMA 5. Let  $(n, m) \in \mathbb{A}_N[P, P]$ . Then  $(n, m) \notin \mathbb{A}_{N+1}[P, P]$  if and only if  $m = \mathfrak{b}(N)$ . Since  $\mathfrak{b}(N) \in \mathcal{C}_{M+1} \setminus \mathcal{C}_M$ , where  $M = \lfloor N^{1/d} \rfloor$  and  $\mathcal{C}_M$  has been introduced in (B.19), there exists  $j \in \{1, \ldots, d\}$  such that  $\mathfrak{b}(N)_j = M$ ; let  $j_*$  be the smallest j satisfying this property. Notice also  $n \in \mathcal{C}_{M+1}$  and  $m \in \mathbb{N}^d \setminus \mathcal{C}_M$ . As soon as  $M > \max(k_1, \ldots, k_d)$ , that we assume from now, the equality  $m = \mathfrak{b}(N)$  can only happen if  $j_* \notin S \setminus P$ . Indeed, if  $j_* \in S \setminus P$ , then  $m_{j_*} = |n_{j_*} - k_{j_*}| \le \max(n_{j_*} - 1, k_{j_*}) \le M - 1$ . As a consequence,

$$\left|\mathbb{A}_{N}[P,P]\right| \leq \left|\mathbb{A}_{N+1}[P,P]\right| \quad \text{if } j_{*} \in S \setminus P.$$

Next, assume that  $j_* \in P$  or  $j_* \notin S$ . We claim that if we set

(B.30) 
$$\tilde{m}_j = \begin{cases} m_j + k_j & \text{if } j \in P, \\ |m_j - k_j| & \text{if } j \in S \setminus P, \\ m_j & \text{if } j \notin S, \end{cases}$$

then  $(\boldsymbol{m}, \tilde{\boldsymbol{m}}) \in \mathbb{A}_{N+1}[P, P] \setminus \mathbb{A}_N[P, P]$ . This would show in particular that

 $|\mathbb{A}_N[P, P]| \le |\mathbb{A}_{N+1}[P, P]|$  if either  $j_* \in P$  or  $j_* \notin S$ ,

and thus complete the proof of (a). That  $(\boldsymbol{m}, \tilde{\boldsymbol{m}}) \in \mathbb{A}_{N+1}[P, P] \setminus \mathbb{A}_N[P, P]$  is by construction obvious provided one can show  $\tilde{\boldsymbol{m}} \ge \mathfrak{b}(N+1)$ .

If  $j_* \in P$ , then we have

$$\tilde{m}_{j_*} = m_{j_*} + k_{j_*} = M + k_{j_*} > M$$

and thus  $\tilde{\boldsymbol{m}} \in \mathbb{N}^d \setminus \mathcal{C}_{M+1}$ . As a consequence, there exists  $m \ge (M+1)^d$  such that  $\tilde{\boldsymbol{m}} = \mathfrak{b}(m)$  and, since  $N+1 \le (M+1)^d$ , we have shown  $\tilde{\boldsymbol{m}} \ge \mathfrak{b}(N+1)$ .

If  $j_* \notin S$ , we argue by contradiction and assume  $\tilde{m} \leq \mathfrak{b}(N) = m$ . We shall see this is not compatible with the graded lexicographic order. Indeed, since by construction  $\tilde{m} \neq m$ and  $n \neq m$  (because  $k \neq (0, ..., 0)$  by assumption), we actually have  $\tilde{m} < m$  and n < m. Because  $j_* \notin S$  by assumption, we moreover have  $n_{j_*} = m_{j_*} = \tilde{m}_{j_*} = M$ , and thus  $n, m, \tilde{m} \in C_{M+1} \setminus C_M$ . As a consequence,  $n <_{\text{lex}} m$  and  $\tilde{m} <_{\text{lex}} m$  in the lexicographic order. This means there exists  $\gamma \in \{1, ..., d\}$  such that  $n_i = m_i$  for every  $i < \gamma$  and  $n_{\gamma} < m_{\gamma}$ , and equivalently  $i \notin S$  when  $i < \gamma$  and  $\gamma \in P$ . Similarly, there exists  $\eta \in \{1, ..., d\}$  such that  $\tilde{m}_i = m_i$  for every  $i < \eta$  and  $\tilde{m}_{\eta} < m_{\eta}$ , and thus  $i \notin S$  when  $i < \eta$  and  $\eta \in S \setminus P$ . But this is impossible, and thus  $\tilde{m} \ge \mathfrak{b}(N + 1)$ , which completes the proof of (a).

Part (b) is proved by following the exact same line of arguments; in this setting one should also check that if  $(n, m) \in \mathbb{A}_N[P, P; \alpha]$  then  $m_\alpha \leq k_\alpha$  in order to show  $(m, \tilde{m})$  (with  $\tilde{m}$ defined in (B.30)) actually belongs to  $\mathbb{A}_{N+1}[P, P; \alpha]$ . Recalling  $\alpha \in S \setminus P$  by assumption this is clear, indeed  $m_\alpha = |n_\alpha - k_\alpha|$  together with  $n_\alpha \leq k_\alpha$  yield  $m_\alpha = k_\alpha - n_\alpha \leq k_\alpha$ .  $\Box$  PROOF OF LEMMA 6. To prove (a), fix  $P \subset S$  and assume  $M > \max(k_1, \ldots, k_d)$ . It follows from the definitions (B.13), (B.22) that

(B.31)  
$$\mathbb{A}_{M^{d}}^{*}[P] = \left\{ (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{M^{d}} \middle| \begin{array}{l} n_{j} + k_{j} = m_{j}, \\ n_{j} - k_{j} = m_{j}, \\ n_{j} = m_{j}, \end{array} \right.$$
$$\left. \begin{array}{l} n_{j} \neq 0, & \text{if } j \in P \\ n_{j} > k_{j}, & \text{if } j \in S \setminus P \\ n_{j} > k_{j}, & \text{if } j \notin S \end{array} \right\}.$$

Recall that  $\mathbb{A}_{M^d} = \mathcal{C}_M \times (\mathbb{N}^d \setminus \mathcal{C}_M)$  where  $\mathcal{C}_M$  is defined in (B.19). Clearly, if we set

(B.32)  $C_M[P] = \{ \boldsymbol{n} \in C_M : \text{there exists } \boldsymbol{m} \in \mathbb{N}^d \setminus C_M, (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{M^d}^*[P] \}$ 

then  $(n, m) \mapsto n$  is a bijection from  $\mathbb{A}^*_{M^d}[P]$  to  $\mathcal{C}_M[P]$ . We claim that if for any  $p \in P$  we set

(B.33) 
$$\mathcal{C}_{M}^{(p)}[P] = \left\{ \boldsymbol{n} \in \mathcal{C}_{M} \mid \begin{array}{c} 1 \leq n_{j} \leq M-1 \text{ if } j \in P \\ k_{j} < n_{j} \leq M-1 \text{ if } j \in S \setminus P \end{array} \right\},$$

then we have

(B.34) 
$$\mathcal{C}_M[P] = \bigcup_{p \in P} \mathcal{C}_M^{(p)}[P].$$

Indeed, let  $\mathbf{n} \in \mathcal{C}_M[P]$ . By definition there exists  $\mathbf{m} \in \mathbb{N}^d \setminus \mathcal{C}_M$  such that  $(\mathbf{n}, \mathbf{m}) \in \mathbb{A}_{M^d}^*[P]$ . This provides (see (B.31)) that  $1 \le n_j \le M - 1$  if  $j \in P$ , that  $k_j < n_j \le M - 1$  if  $j \in S \setminus P$ , and the existence of  $p \in \{1, \ldots, d\}$  satisfying  $m_p \ge M$ . Since  $\mathbf{n} \in \mathcal{C}_M$  then  $n_p \le M - 1$ , and thus  $p \in P$  because otherwise  $m_p \le n_p$ . Together with the equation  $n_p + k_p = m_p$  this finally yields that  $M - k_p \le n_p \le M - 1$ , namely  $\mathbf{n} \in \mathcal{C}_M^{(p)}[P]$  for some  $p \in P$ . As for the reverse inclusion, if  $\mathbf{n} \in \mathcal{C}_M^{(\alpha)}[P]$  for some  $\alpha \in P$  then set

$$m_j = \begin{cases} n_j + k_j & \text{if } j \in P, \\ n_j - k_j & \text{if } j \in S \setminus P, \\ n_j & \text{if } j \notin S, \end{cases}$$

and observe that  $\boldsymbol{m} \in \mathbb{N}^d \setminus \mathcal{C}_M$  since  $m_p \ge M$  and  $n_j - k_j \ge 0$  for every  $j \in S \setminus P$ . Thus, since clearly  $(\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}^*_{M^d}[P]$ , we have shown  $\boldsymbol{n} \in \mathcal{C}_M[P]$  and (B.34) is proved.

Next, since for every distinct  $p_1, \ldots, p_q \in P$  the definition (B.33) yields

$$|\mathcal{C}_M^{(p_1)}[P] \cap \cdots \cap \mathcal{C}_M^{(p_q)}[P]| = k_{p_1} \cdots k_{p_q} M^{d-q} + \mathcal{O}(M^{d-q-1}),$$

then (a) follows from (B.34) and the inclusion-exclusion principle.

We now turn to (b) and fix  $\alpha \in S \setminus P$ . Let  $\mathcal{C}_M^{(d)}$  be the *d*-dimensional discrete hypercube of length *M* defined as in (B.19). We then set

$$\mathbb{A}_{M^{d-1}}^{(d-1)} = \mathcal{C}_M^{(d-1)} \times (\mathbb{N}^{d-1} \setminus \mathcal{C}_M^{(d-1)})$$

and introduce

$$\mathbb{A}_{M^{d-1}}^{*}[P]^{(d-1)} = \left\{ (n, m) \in \mathbb{A}_{M^{d-1}}^{(d-1)} \middle| \begin{array}{l} n_{j} + k_{j} = m_{j}, \\ n_{j} - k_{j} = m_{j}, \\ n_{j} = m_{j}, \end{array} \right.$$
$$\left. \begin{array}{l} n_{j} \neq 0, & \text{if } j \in P \\ n_{j} > k_{j}, & \text{if } j \in S \setminus (P \cup \{\alpha\}) \\ \text{if } j \notin S \setminus \{\alpha\} \end{array} \right\}.$$

The statement (a) of the lemma applied in dimension d - 1 then provides

(B.35) 
$$|\mathbb{A}_{M^{d-1}}^*[P]^{(d-1)}| = o(M^{d-1}).$$

Consider the map  $\mathfrak{p}: \mathbb{N}^d \to \mathbb{N}^{d-1}$  defined by

 $\mathfrak{p}(n_1,\ldots,n_d)=(n_1,\ldots,n_{\alpha-1},n_{\alpha+1},\ldots,n_d).$ 

Let  $(\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{M^d}^*[P; \alpha]$ . Since  $\boldsymbol{m} \in \mathbb{N}^d \setminus \mathcal{C}_M^{(d)}$  and  $m_\alpha \leq k_\alpha < M$ , there exists  $j \neq \alpha$  such that  $m_j \geq M$ . It follows that  $(\mathfrak{p}(\boldsymbol{n}), \mathfrak{p}(\boldsymbol{m})) \in \mathcal{C}_M^{(d-1)} \times (\mathbb{N}^{d-1} \setminus \mathcal{C}_M^{(d-1)})$ , and thus  $(\mathfrak{p}(\boldsymbol{n}), \mathfrak{p}(\boldsymbol{m})) \in \mathbb{A}_{M^{d-1}}^*[P]^{(d-1)}$ . As a consequence, we have the upper bound

$$|A_{M^d}^*[P;\alpha]| \le k_\alpha |(\mathfrak{p} \times \mathfrak{p})(A_{M^d}^*[P;\alpha])| \le k_\alpha |A_{M^{d-1}}^*[P]^{(d-1)}|,$$

and thus (b) follows from (B.35).  $\Box$ 

**B.3.** Covariance asymptotics: The general case. We now extend Proposition 4 to the general setting of measures satisfying the assumptions of Theorem 2.1. More precisely, we prove the following.

PROPOSITION 5. Let  $\mu = \mu_1 \otimes \cdots \otimes \mu_d$ , where the  $\mu_j$ 's are Nevai-class probability measures on I. Let  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  and  $\mathbf{x}_1^*, \ldots, \mathbf{x}_N^*$  be random variables drawn from the multivariate OP Ensembles with respective reference measures  $\mu$  and  $\mu_{eq}^{\otimes d}$ . Then, given any polynomial functions P, Q on  $\mathbb{R}^d$ ,

(B.36)  
$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \left| \mathbb{C}ov\left[\sum_{i=1}^{N} P(\mathbf{x}_i), \sum_{j=1}^{N} Q(\mathbf{x}_i)\right] - \mathbb{C}ov\left[\sum_{i=1}^{N} P(\mathbf{x}_i^*), \sum_{j=1}^{N} Q(\mathbf{x}_i^*)\right] \right| = 0.$$

For the proof of the proposition, we use a few ingredients from the Step 2 of the proof of Lemma 1 to which we refer the reader to.

PROOF OF PROPOSITION 5. By linearity, it is enough to prove the proposition with  $P(x) = x_1^{\alpha_1} \cdots x_d^{\alpha_d}$  and  $Q(x) = x_1^{\beta_1} \cdots x_d^{\beta_d}$  for any fixed  $\alpha_1, \beta_1, \dots, \alpha_d, \beta_d \in \mathbb{N}$ . Lemma 4 then provides

$$(B.37) \qquad \qquad \mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} P(\mathbf{x}_{i}), \sum_{j=1}^{N} Q(\mathbf{x}_{i})\right]$$
$$= \sum_{n=0}^{N-1} \sum_{m=N}^{\infty} \langle x_{1}^{\alpha_{1}} \cdots x_{d}^{\alpha_{d}} \varphi_{n}, \varphi_{m} \rangle_{L^{2}(\mu)} \langle x_{1}^{\beta_{1}} \cdots x_{d}^{\beta_{d}} \varphi_{n}, \varphi_{m} \rangle_{L^{2}(\mu)}$$
$$= \sum_{(\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{N}} \prod_{j=1}^{d} \langle x^{\alpha_{j}} \varphi_{n_{j}}^{(j)}, \varphi_{m_{j}}^{(j)} \rangle_{L^{2}(\mu_{j})} \langle x^{\beta_{j}} \varphi_{n_{j}}^{(j)}, \varphi_{m_{j}}^{(j)} \rangle_{L^{2}(\mu_{j})},$$

where we recall that

$$\mathbb{A}_N = \left\{ \left( \mathfrak{b}(n), \mathfrak{b}(m) \right) : n \le N - 1, m \ge N \right\} \subset \mathbb{N}^d \times \mathbb{N}^d.$$

In particular, by choosing  $\mu = \mu_{eq}^{\otimes d}$  in (B.37), we obtain

(B.38)  

$$\mathbb{C}\operatorname{ov}\left[\sum_{i=1}^{N} P(\mathbf{x}_{i}^{*}), \sum_{j=1}^{N} Q(\mathbf{x}_{i}^{*})\right]$$

$$= \sum_{(\boldsymbol{n},\boldsymbol{m})\in\mathbb{A}_{N}} \prod_{j=1}^{d} \langle x^{\alpha_{j}} T_{n_{j}}, T_{m_{j}} \rangle_{L^{2}(\mu_{eq})} \langle x^{\beta_{j}} T_{n_{j}}, T_{m_{j}} \rangle_{L^{2}(\mu_{eq})}$$

Thus, by combining (B.37) and (B.38), we see that, if we set for convenience

(B.39)  
$$E(\boldsymbol{n}, \boldsymbol{m}) = \left| \prod_{j=1}^{d} \langle x^{\alpha_{j}} \varphi_{n_{j}}^{(j)}, \varphi_{m_{j}}^{(j)} \rangle_{L^{2}(\mu_{j})} \langle x^{\beta_{j}} \varphi_{n_{j}}^{(j)}, \varphi_{m_{j}}^{(j)} \rangle_{L^{2}(\mu_{j})} - \prod_{j=1}^{d} \langle x^{\alpha_{j}} T_{n_{j}}, T_{m_{j}} \rangle_{L^{2}(\mu_{eq})} \langle x^{\beta_{j}} T_{n_{j}}, T_{m_{j}} \rangle_{L^{2}(\mu_{eq})} \right|,$$

then proving the proposition amounts to showing that

(B.40) 
$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \sum_{(\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N} E(\boldsymbol{n}, \boldsymbol{m}) = 0.$$

Next, for every  $j \in \{1, ..., d\}$ , the three-term recurrence relation reads

(B.41) 
$$x\varphi_n^{(j)} = a_n^{(j)}\varphi_{n+1}^{(j)} + b_n^{(j)}\varphi_n^{(j)} + a_{n-1}^{(j)}\varphi_{n-1}^{(j)}, \quad n \ge 0,$$

where we set  $a_{-1}^{(j)} = 0$ . As in Step 2 of the proof of Lemma 1, we complete the sequences of recurrence coefficients  $(a_n^{(j)})_{n \in \mathbb{N}}$  and  $(b_n^{(j)})_{n \in \mathbb{N}}$  introduced into sequences  $(a_n^{(j)})_{n \in \mathbb{Z}}, (b_n^{(j)})_{n \in \mathbb{Z}}$ , where we set  $a_n^{(j)} = b_n^{(j)} = 0$  for every n < 0. We thus obtain the representations

(B.42) 
$$\{x^{\alpha}\varphi_{n_{j}}^{(j)},\varphi_{m_{j}}^{(j)}\}_{L^{2}(\mu_{j})} = \mathbf{1}_{|n_{j}-m_{j}| \leq \alpha} \sum_{\gamma:(0,n_{j}-m_{j})\to(\alpha,0)} \prod_{e\in\gamma} \omega(e)_{\{(a_{n+m_{j}}^{(j)}),(b_{n+m_{j}}^{(j)})\}},$$

and

(B.43)  $\langle x^{\alpha} T_{n_j}, T_{m_j} \rangle_{L^2(\mu_{eq})}$  $= \mathbf{1}_{|n_j - m_j| \le \alpha} \sum_{\gamma: (0, n_j - m_j) \to (\alpha, 0)} \prod_{e \in \gamma} \omega(e)_{\{(a_{n+m_j}^*), (b_{n+m_j}^*)\}},$ 

where we recall that w(e) was introduced in (A.18). Since the measures  $\mu_j$  are Nevaiclass by assumption, we have  $a_n^{(j)} - a_n^* \to 0$  and  $b_n^{(j)} - b_n^* \to 0$  as  $n \to \infty$  for every  $j \in \{1, \ldots, d\}$ . Notice that for every  $n_j$ , the right-hand side of (B.42) is a polynomial function of  $a_{m_j-\alpha}^{(j)}, b_{m_j-\alpha}^{(j)}, \ldots, a_{m_j+\alpha}^{(j)}, b_{m_j+\alpha}^{(j)}$  and does not depend on any other recurrence coefficients. Thus, we obtain for every fixed  $\alpha \in \mathbb{N}$ ,

(B.44) 
$$\sup_{n_j \in \mathbb{N}} |\langle x^{\alpha} \varphi_{n_j}^{(j)}, \varphi_{m_j}^{(j)} \rangle_{L^2(\mu_j)} - \langle x^{\alpha} T_{n_j}, T_{m_j} \rangle_{L^2(\mu_{eq})}| \xrightarrow[m_j \to \infty]{} 0.$$

Moreover, we see from (B.39), (B.42) and (B.43) that E(n, m) = 0 except when  $|n_j - m_j| \le \min(\alpha_j, \beta_j)$  for every  $j \in \{1, ..., d\}$ . We then split the set of contributing indices into

two subsets,

$$\mathbb{A}_N^* = \{ (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N : |n_j - m_j| \le \min(\alpha_j, \beta_j) \} \cap \{ m_j \ge N^{1/2d} \text{ for every } j \}.$$
$$\mathbb{A}_N^0 = \{ (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N : |n_j - m_j| \le \min(\alpha_j, \beta_j) \}$$
$$\cap \{ m_j < N^{1/2d} \text{ for at least one } j \}.$$

It then follows from (B.44) that

(B.45) 
$$\lim_{N \to \infty} \sup_{(\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N^*} |E(\boldsymbol{n}, \boldsymbol{m})| = 0$$

and that there exists C > 0 independent on N satisfying

(B.46) 
$$\sup_{(\boldsymbol{n},\boldsymbol{m})\in\mathbb{A}_N^0} |E(\boldsymbol{n},\boldsymbol{m})| \leq \sup_{(\boldsymbol{n},\boldsymbol{m})\in\mathbb{A}_N} |E(\boldsymbol{n},\boldsymbol{m})| \leq C$$

Next, we write

(B.47) 
$$\frac{1}{N^{1-1/d}} \sum_{(n,m)\in\mathbb{A}_N} E(n,m)$$
$$= \frac{1}{N^{1-1/d}} \sum_{(n,m)\in\mathbb{A}_N^*} E(n,m) + \frac{1}{N^{1-1/d}} \sum_{(n,m)\in\mathbb{A}_N^0} E(n,m)$$
$$\leq \frac{|\mathbb{A}_N^*|}{N^{1-1/d}} \sup_{(n,m)\in\mathbb{A}_N^*} |E(n,m)| + \frac{|\mathbb{A}_N^0|}{N^{1-1/d}} \sup_{(n,m)\in\mathbb{A}_N^0} |E(n,m)|$$

and claim that we have

(B.48) 
$$\limsup_{N \to \infty} \frac{|\mathbb{A}_N^*|}{N^{1-1/d}} < \infty,$$

and, moreover,

(B.49) 
$$\lim_{N \to \infty} \frac{|\mathbb{A}_N^0|}{N^{1-1/d}} = 0.$$

Together with (B.45)–(B.46), this would prove (B.40), and thus the proposition.

We finally prove (B.48) and (B.49) in order to complete the proof of the proposition. Let us set  $\kappa_j = \max(\alpha_j, \beta_j)$  for convenience. Clearly,

(B.50)  
$$|\mathbb{A}_{N}^{*}| = \left| \bigcup_{\boldsymbol{n} \in \mathbb{N}^{d}} \{ \boldsymbol{m} \in \mathbb{N}^{d} : (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{N}^{*} \} \right|$$
$$\leq \max_{\boldsymbol{n} \in \mathbb{N}^{d}} \left| \{ \boldsymbol{m} \in \mathbb{N}^{d} : (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{N}^{*} \} \right|$$
$$\times \left| \{ \boldsymbol{n} \in \mathbb{N}^{d} : (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_{N}^{*} \text{ for some } \boldsymbol{m} \in \mathbb{N}^{d} \} \right|$$

First, since  $|n_j - m_j| \le \kappa_j$  for every *j* as soon as  $(n, m) \in \mathbb{A}_N^*$ , we have the upper bound

(B.51) 
$$\max_{\boldsymbol{n}\in\mathbb{N}^d} |\{\boldsymbol{m}\in\mathbb{N}^d: (\boldsymbol{n},\boldsymbol{m})\in\mathbb{A}_N^*\}| \leq \prod_{j=1}^d (2\kappa_j+1).$$

Next, set  $M = \lfloor N^{1/d} \rfloor$  so that  $M^d \leq N < (M+1)^d$ . If  $(n, m) \in \mathbb{A}_N^*$ , then it satisfies  $n \in \mathcal{C}_{M+1}$ and  $m \in \mathbb{N}^d \setminus \mathcal{C}_M$ , where  $\mathcal{C}_M$  has been introduced in (B.19). Namely, it holds that  $0 \leq n_j \leq M$ for every j and there exists  $j_0$  such that  $m_{j_0} \geq M$ . Together with  $|n_{j_0} - m_{j_0}| \leq \kappa_{j_0}$ , this yields  $M - \kappa_{j_0} \le n_{j_0} \le M$ , and thus provides the upper bound

(B.52) 
$$\left|\left\{\boldsymbol{n} \in \mathbb{N}^d : (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N^* \text{ for some } \boldsymbol{m} \in \mathbb{N}^d\right\}\right| \le \left(\max_{j_0=1}^d \kappa_{j_0} + 1\right) (M+1)^{d-1}$$

By combining (B.50)–(B.52), we have proved (B.48). The proof of (B.49) is similar. More precisely, the only difference is that if  $(n, m) \in \mathbb{A}_N^{(0)}$ , then there exists  $j_1$  such that  $m_{j_1} < \sqrt{N^{\frac{1}{d}}} < \sqrt{M+1}$ . Notice that necessarily  $j_1 \neq j_0$ . Using moreover that  $|n_{j_1} - m_{j_1}| \le \kappa_{j_1}$ , we obtain the upper bound

$$\{\boldsymbol{n} \in \mathbb{N}^d : (\boldsymbol{n}, \boldsymbol{m}) \in \mathbb{A}_N^{(0)} \text{ for some } \boldsymbol{m} \in \mathbb{N}^d\} |$$
  
$$\leq \Big( \max_{\substack{j_0=1\\j_0 \neq j_1}} \kappa_{j_0} + 1 \Big) (\kappa_{j_1} + \sqrt{M+1}) (M+1)^{d-2}$$

in place of (B.52), and (B.49) follows.  $\Box$ 

**B.4. Extension to**  $\mathscr{C}^1$  **functions and conclusion.** We consider a reference measure  $\mu$  satisfying the assumptions of Theorem 2.1 and let  $\mathbf{x}_1, \ldots, \mathbf{x}_N$  be the associated multivariate OP Ensemble. For any *d*-multivariate polynomial *P*, we can write  $P = \sum_{k \in \mathbb{N}^d} \hat{P}(k)T_k$ , where the latter sum is finite. As a consequence of Propositions 4 and 5, we then obtain

(B.53)  
$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V} \operatorname{ar} \left[ \sum_{i=1}^{N} P(\mathbf{x}_i) \right]$$
$$= \sum_{\boldsymbol{k}, \boldsymbol{\ell} \in \mathbb{N}^d} \hat{P}(\boldsymbol{k}) \hat{P}(\boldsymbol{\ell}) \lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{C} \operatorname{ov} \left[ \sum_{i=1}^{N} T_{\boldsymbol{k}}(\mathbf{x}_i), \sum_{i=1}^{N} T_{\boldsymbol{\ell}}(\mathbf{x}_i) \right]$$
$$= \frac{1}{2} \sum_{\boldsymbol{k} = (k_1, \dots, k_d) \in \mathbb{N}^d} (k_1 + \dots + k_d) \hat{P}(\boldsymbol{k})^2 = \sigma_P^2.$$

Therefore, we have proven Proposition 3 provided we restrict the test functions to polynomials. We finally extend this result to  $\mathscr{C}^1$  test functions, and thus complete the proof of this proposition, by means of a density argument.

First, a standard computation yields

(B.54) 
$$\mathbb{V}\mathrm{ar}\left[\sum_{i=1}^{N} f(\mathbf{x}_{i})\right] = \frac{1}{2} \iint (f(x) - f(y))^{2} K_{N}(x, y)^{2} \mu(\mathrm{d}x) \mu(\mathrm{d}y).$$

This indeed follows from (2.1)–(2.3) with k = 1, 2, and that  $K_N(x, y)$  is a symmetric reproducing kernel.

Now, for any  $f \in \mathscr{C}^1(I^d, \mathbb{R})$ , we set

(B.55) 
$$||f||_{\text{Lip}} = \sup_{x \in I^d} ||\nabla f(x)||,$$

so that  $|f(x) - f(y)| \le ||f||_{\text{Lip}} ||x - y||$  for every  $x \ne y$ . If we consider the monomials defined by

$$(B.56) e_j(x_1,\ldots,x_d) = x_j,$$

then formula (B.54) yields

$$\mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} f(\mathbf{x}_{i})\right] = \frac{1}{2} \iint \left(f(x) - f(y)\right)^{2} K_{N}(x, y)^{2} \mu(\mathrm{d}x) \mu(\mathrm{d}y)$$

$$\leq \|f\|_{\text{Lip}}^{2} \sum_{j=1}^{d} \frac{1}{2} \iint (e_{j}(x) - e_{j}(y))^{2} K_{N}(x, y)^{2} \mu(dx) \mu(dy)$$
$$= \|f\|_{\text{Lip}}^{2} \sum_{j=1}^{d} \mathbb{V} \operatorname{ar} \left[ \sum_{i=1}^{N} e_{j}(\mathbf{x}_{i}) \right]$$

and, as a consequence of (B.53),

(B.57) 
$$\limsup_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}ar\left[\sum_{i=1}^{N} f(\mathbf{x}_i)\right] \le C \|f\|_{\text{Lip}}^2, \quad C = \sum_{j=1}^{d} \sigma_{e_j}^2.$$

Proposition 1 also provides the upper bound

(B.58) 
$$\sigma_f^2 \le \frac{1}{2} \|f\|_{\text{Lip}}^2$$

Next, Theorem 5 of [62] yields the existence of a sequence of multivariate polynomials  $(P_{\varepsilon})_{\varepsilon>0}$  such that  $||P_{\varepsilon} - f||_{\text{Lip}} \le \varepsilon$ , and hence

(B.59) 
$$\limsup_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V} \operatorname{ar} \left[ \sum_{i=1}^{N} f(\mathbf{x}_{i}) - \sum_{i=1}^{N} P_{\varepsilon}(\mathbf{x}_{i}) \right] \leq C \varepsilon^{2}, \quad \text{and}$$
$$\sigma_{f-P_{\varepsilon}}^{2} \leq \frac{\varepsilon^{2}}{2}.$$

Since  $(X, Y) \mapsto \mathbb{C}ov(X, Y)$  is a symmetric positive bilinear form, it satisfies the Cauchy– Schwarz inequality, and thus the triangle inequality  $\mathbb{V}ar(X+Y)^{1/2} \leq \mathbb{V}ar(X)^{1/2} + \mathbb{V}ar(Y)^{1/2}$ holds true, which in turn yields the inequality

(B.60) 
$$|\operatorname{Var}(X)^{1/2} - \operatorname{Var}(Y)^{1/2}| \le \operatorname{Var}(X - Y)^{1/2}.$$

For the same reason, the limiting variance satisfies  $|\sigma_f - \sigma_g| \le \sigma_{f-g}$ . As a consequence, by taking  $X = \sum f(\mathbf{x}_i)$  and  $Y = \sum P_{\varepsilon}(\mathbf{x}_i)$  in (B.60), and using these two inequalities together with (B.53) and (B.59), we obtain by letting  $N \to \infty$  and then  $\varepsilon \to 0$  that

$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} f(\mathbf{x}_{i})\right] = \lim_{\varepsilon \to 0} \lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} P_{\varepsilon}(\mathbf{x}_{i})\right]$$
$$= \lim_{\varepsilon \to 0} \sigma_{P_{\varepsilon}}^{2}$$
$$= \sigma_{f}^{2}$$

and the proof of Proposition 3 is therefore complete.

#### APPENDIX C: MONTE CARLO WITH DPPS: PROOF OF THEOREM 2.2

The aim of this section is to prove the following variance decay.

**PROPOSITION 6.** Assume  $\mu(dx) = \omega(x) dx$  with  $\omega$  positive and  $\mathcal{C}^1$  on  $(-1, 1)^d$ . Assume further that  $\mu$  satisfies Assumption 1. For every  $f \in \mathcal{C}$ , we have

$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}\operatorname{ar}\left[\sum_{i=1}^{N} \frac{Nf(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} - \sum_{i=1}^{N} \frac{\omega(x)f(\mathbf{x}_i)}{\omega_{\text{eq}}^{\otimes d}(\mathbf{x}_i)}\right] = 0.$$

Before proving Proposition 6, we argue that it implies Theorem 2.2. Indeed, (B.60) then implies that

$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}\mathrm{ar} \left[ \sum_{i=1}^{N} \frac{Nf(\mathbf{x}_i)}{K_N(\mathbf{x}_i, \mathbf{x}_i)} \right] = \lim_{N \to \infty} \frac{1}{N^{1-1/d}} \mathbb{V}\mathrm{ar} \left[ \sum_{i=1}^{N} \frac{\omega(x) f(\mathbf{x}_i)}{\omega_{\mathrm{eq}}^{\otimes d}(\mathbf{x}_i)} \right]$$
$$= \Omega_{f,\omega}^2,$$

the last equality following from Theorem 2.1. Now Theorem A.3 applies with  $f_N(x) = Nf(x)/K_N(x, x)$  to yield Theorem 2.2.

From now on, we fix  $f \in \mathscr{C}$ . It is thus a  $\mathscr{C}^1$  function and there exists  $\varepsilon > 0$  so that  $\operatorname{Supp}(f) \subset I_{\varepsilon}^d$ . If we set for convenience

$$E_N(x) = \frac{N}{K_N(x,x)} - \sum_{i=1}^N \frac{\omega(x)}{\omega_{\text{eq}}^{\otimes d}(x)}, \quad x \in I^d,$$

then Theorem A.4 yields  $||f E_N||_{\infty} = \sup_{I^d} |f E_N| \to 0$  as  $N \to \infty$ .

In order to prove Proposition 6, we start with the formula coming from (B.54),

$$\operatorname{Var}\left[\sum_{i=1}^{N} f(\mathbf{x}_{i}) E_{N}(\mathbf{x}_{i})\right] = \frac{1}{2} \iint \left(f E_{N}(x) - f E_{N}(y)\right)^{2} K_{N}(x, y)^{2} \mu(\mathrm{d}x) \mu(\mathrm{d}y)$$

and split the integral in several terms that we shall analyse separately.

**C.1. The off-diagonal contribution.** Given any  $\delta > 0$ , we first consider the contribution

(C.1) 
$$\frac{1}{2} \iint_{\|x-y\|>\delta} (f E_N(x) - f E_N(y))^2 K_N(x, y)^2 \mu(dx) \mu(dy)$$

By rough estimates, we obtain

$$(C.1) \leq \|f E_N\|_{\infty}^2 \iint_{\|x-y\|>\delta} K_N(x, y)^2 \mu(dx) \mu(dy)$$
  
$$\leq \frac{1}{\delta^2} \|f E_N\|_{\infty}^2 \sum_{j=1}^d \iint (x_j - y_j)^2 K_N(x, y)^2 \mu(dx) \mu(dy)$$
  
$$\leq \frac{2}{\delta^2} \|f E_N\|_{\infty}^2 \sum_{j=1}^d \mathbb{V} \operatorname{ar} \left[ \sum_{i=1}^N e_j(\mathbf{x}_i) \right],$$

where the monomials  $e_j$  were defined in (B.56). As a consequence, using Proposition 3 and that  $||f E_N||_{\infty} \to 0$  as  $N \to \infty$ , we get

(C.2) 
$$\lim_{N \to \infty} \frac{1}{N^{1-1/d}} \iint_{\|x-y\| > \delta} (f E_N(x) - f E_N(y))^2 \\ \times K_N(x, y)^2 \mu(dx) \mu(dy) = 0$$

for every  $\delta > 0$ .

C.2. The diagonal contribution. By (C.2), it is sufficient to show

(C.3)  
$$\limsup_{\delta \to 0} \limsup_{N \to \infty} \frac{1}{N^{1-1/d}} \iint_{\|x-y\| \le \delta} (f E_N(x) - f E_N(y))^2 \times K_N(x, y)^2 \mu(\mathrm{d}x) \mu(\mathrm{d}y) = 0$$

in order to complete the proof of Proposition 6.

Set for convenience

(C.4) 
$$\mathscr{D}_g(x, y) = \frac{g(x) - g(y)}{\|x - y\|}, \quad x, y \in I^d,$$

so that,  $|\mathscr{D}_g(x, y)| \le ||g||_{\text{Lip}}$ . For every  $\delta > 0$  small enough, we have for any x, y satisfying  $||x - y|| \le \delta$ ,

$$\mathcal{D}_{fE_N}(x, y)^2 = \left( \mathcal{D}_f(x, y) E_N(x) + \mathcal{D}_{E_N}(x, y) f(y) \right)^2$$
  

$$\leq 2 \mathcal{D}_f(x, y)^2 E_N(x)^2 + 2 \mathcal{D}_{E_N}(x, y)^2 f(y)^2$$
  

$$\leq 2 \|f\|_{\text{Lip}}^2 \|\mathbf{1}_{I_{\varepsilon/2}^d}^d E_N\|_{\infty}^2 + 2 \|f\|_{\infty}^2 \mathcal{D}_{E_N}(x, y)^2 \mathbf{1}_{I_{\varepsilon/2}^d \times I_{\varepsilon/2}^d}^d (x, y).$$

Indeed, notice that if  $x \in I_{\varepsilon}^{d}$  or  $y \in I_{\varepsilon}^{d}$ , then  $x, y \in I_{\varepsilon/2}^{d}$  for every  $\delta > 0$  small enough. Since f is by assumption supported on  $I_{\varepsilon}^{d}$ , we know that  $\mathscr{D}_{fE_{N}}(x, y)$  vanishes outside of  $I_{\varepsilon/2}^{d} \times I_{\varepsilon/2}^{d}$ . This is the reason for the presence of  $\mathbf{1}_{I_{\varepsilon/2}^{d}}$  in the last inequality.

With the notation  $Q_N$  introduced in (A.31), we thus obtain

(C.5)  

$$\frac{1}{2N^{1-1/d}} \iint_{\|x-y\| \le \delta} (fE_N(x) - fE_N(y))^2 K_N(x, y)^2 \mu(dx) \mu(dy) \\
= \frac{1}{2} \iint_{\|x-y\| \le \delta} \mathscr{D}_{fE_N}(x, y)^2 Q_N(dx, dy) \\
\le \|f\|_{\text{Lip}}^2 \|\mathbf{1}_{I_{\varepsilon/2}^d} E_N\|_{\infty}^2 \iint_{\|x-y\| \le \delta} Q_N(dx, dy) \\
+ \|f\|_{\infty}^2 \iint_{I_{\varepsilon/2}^d \times I_{\varepsilon/2}^d, \|x-y\| \le \delta} \mathscr{D}_{E_N}(x, y)^2 Q_N(dx, dy).$$

Moreover, because  $\omega$  is  $\mathscr{C}^1$  on  $I^d_{\varepsilon/2}$  by assumption, and because  $\omega_{eq}^{\otimes d}$  is also  $\mathscr{C}^1$  and positive there, one similarly has, for every  $x, y \in I^d_{\varepsilon/2}$ ,

(C.6)  
$$\mathcal{D}_{E_N}(x, y)^2 \leq 2\mathcal{D}_N(x, y)^2 + 2\mathcal{D}_{\omega/\omega_{eq}^{\otimes d}}(x, y)^2$$
$$\leq 2\mathcal{D}_N(x, y)^2 + 2\|\omega/\omega_{eq}^{\otimes d}\|_{Lip},$$

where  $\mathscr{D}_N$  is defined in (A.30).

Next, we have for every C > 0,

(C.7)  
$$\iint_{I_{\varepsilon/2}^d \times I_{\varepsilon/2}^d, \|x-y\| \le \delta} \mathscr{D}_N(x, y)^2 Q_N(dx, dy)$$
$$\leq C^2 \iint_{\|x-y\| \le \delta} Q_N(dx, dy)$$
$$+ \iint_{I_{\varepsilon/2}^d \times I_{\varepsilon/2}^d, \|x-y\| \le \delta} \mathbf{1}_{|\mathscr{D}_N(x, y)| > C} \mathscr{D}_N(x, y)^2 Q_N(dx, dy)$$

We now make use of the following lemma, the proof of which is deferred to Section C.3.

Lemma 7.

(C.8) 
$$\lim_{\delta \to 0} \limsup_{N \to \infty} \iint_{\|x-y\| \le \delta} Q_N(dx, dy) = 0.$$

As a consequence, (C.5), (C.6) and (C.7) together yield, for every C > 0,

(C.9)  
$$\lim_{\delta \to 0} \sup_{N \to \infty} \frac{1}{2N^{1-1/d}} \iint_{\|x-y\| \le \delta} (f E_N(x) - f E_N(y))^2 \times K_N(x, y)^2 \mu(dx) \mu(dy)$$
$$\le 2\|f\|_{\infty}^2 \limsup_{\delta \to 0} \limsup_{N \to \infty} \iint_{I_{\varepsilon/2}^d \times I_{\varepsilon/2}^d, \|x-y\| \le \delta} \mathscr{D}_N(x, y)^2 \times \mathbf{1}_{|\mathscr{D}_N(x, y)| > C} Q_N(dx, dy).$$

Assumption 1 allows us to conclude the proof of Proposition 6, up to the proof of Lemma 7.

## C.3. Proof of Lemma 7.

PROOF. First,

0)  
$$\iint_{\|x-y\| \le \delta} Q_N(dx, dy) \le \frac{1}{N^{1-1/d}} \sum_{j=1}^d \iint_{|x_j-y_j| \le \delta} (x_j - y_j)^2 K_N(x, y)^2 \mu(dx) \mu(dy).$$

(C.10)

We fix  $j \in \{1, ..., d\}$  and use the notation of the proof of Lemma 2. It comes

$$K_N(x, y) = \sum_{[k] \in \Gamma_N/\sim} K_{N_j([k])+1}^{(j)}(x_j, y_j) \prod_{\alpha \neq j} \varphi_{k_\alpha}(x_\alpha) \varphi_{k_\alpha}(y_\alpha).$$

Squaring, integrating and using the orthonormality relations,

$$\begin{split} \iint_{|x_j - y_j| \le \delta} (x_j - y_j)^2 K_N(x, y)^2 \mu(dx) \mu(dy) \\ &= \sum_{[k], [\ell] \in \Gamma_N / \sim} \mathbf{1}_{\sigma(k) = \sigma(\ell)} \\ &\times \iint_{|x_j - y_j| \le \delta} (x_j - y_j)^2 K_{N_j([k]) + 1}^{(j)}(x_j, y_j) \\ &\times K_{N_j([\ell]) + 1}^{(j)}(x_j, y_j) \mu_j(dx_j) \mu_j(dy_j) \\ &= \sum_{[k] \in \Gamma_N / \sim} \iint_{|x_j - y_j| \le \delta} (x_j - y_j)^2 K_{N_j([k]) + 1}^{(j)}(x_j, y_j)^2 \mu_j(dx_j) \mu_j(dy_j). \end{split}$$

Recall  $M = \lfloor N^{1/d} \rfloor$  and  $C_M \subset \Gamma_N \subset C_{M+1}$ . By definition of  $\mathfrak{b}$ , we have for every  $1 \leq m \leq M-2$ ,

$$|\{[\mathbf{k}] \in \Gamma_N / \sim : N_j([\mathbf{k}]) = m\}| \leq dM^{d-2}.$$

Notice also that (A.13) yields

$$\iint (x_j - y_j)^2 K_m^{(j)}(x_j, y_j)^2 \mu_j(\mathrm{d} x_j) \mu_j(\mathrm{d} y_j) = 2a_m^2,$$

which is bounded for every *m* since  $a_m \rightarrow 1/2$  by assumption. Now

$$(\mathbf{C}.11) = \left[\sum_{\substack{[\mathbf{k}] \in \Gamma_N/\sim \\ N_j([\mathbf{k}]) < \sqrt{M}}} + \sum_{\substack{[\mathbf{k}] \in \Gamma_N/\sim \\ N_j([\mathbf{k}]) \ge \sqrt{M}}}\right] \iint_{|x_j - y_j| \le \delta} (x_j - y_j)^2$$

(C.12)  

$$\times K_{N_{j}([k])+1}^{(j)}(x_{j}, y_{j})^{2} \mu_{j}(dx_{j}) \mu_{j}(dy_{j})$$

$$\leq \mathcal{O}(M^{d-2+1/2})$$

$$+ M^{d-1} \max_{\sqrt{M} \leq m \leq M} \iint_{|x_{j}-y_{j}| \leq \delta} (x_{j}-y_{j})^{2}$$

$$\times K_{m+1}^{(j)}(x_{j}, y_{j})^{2} \mu_{j}(dx_{j}) \mu_{j}(dy_{j}).$$

Moreover, Lemma 1 yields

$$\max_{\sqrt{M} \le m \le M} \iint_{|x_j - y_j| \le \delta} (x_j - y_j)^2 K_{m+1}^{(j)}(x_j, y_j)^2 \mu_j(\mathrm{d}x_j) \mu_j(\mathrm{d}y_j)$$
  
$$\to \iint_{|x-y| \le \delta} L(x, y) \, \mathrm{d}x \, \mathrm{d}y$$

as  $M \to \infty$ . Combined with (C.10)–(C.12), we obtain

$$\limsup_{N \to \infty} \iint_{\|x-y\| \le \delta} Q_N(dx, dy) \le d \iint_{\|x-y\| \le \delta} L(x, y) dx dy.$$

Finally, since *L* is integrable, the lemma follows by letting  $\delta \rightarrow 0$ .  $\Box$ 

The proof of Proposition 6 is therefore complete.

## APPENDIX D: ADDITIONAL EXPERIMENTAL RESULTS

In Figure 5, we display the results of the linear regression in Section 3.3. In Figure 6, we display those of the linear regression in Section 3.4.



FIG. 5. Summary of the importance sampling results.



FIG. 6. Summary of the crude Monte Carlo results for a test function that violates the assumptions of Theorem 2.2.

Acknowledgements. AH would like to thank Bernhard Beckermann, Jeff Geronimo and Kurt Johansson for useful and stimulating discussions. This work started while RB and AH were, respectively, at University of Oxford and KTH Royal Institute of Technology.

Both authors acknowledge support from ANR through grant BoB ANR-16-CE23-0003.

The first author was further supported by EPSRC grant EP/I017909/1 and ANR grant BNPSI ANR-13-BS03-0006.

The second author was further supported by grant KAW 2010.0063 from the Knut and Alice Wallenberg Foundation and Labex CEMPI ANR-11-LABX-0007-01.

#### REFERENCES

- AMEUR, Y., HEDENMALM, H. and MAKAROV, N. (2011). Fluctuations of eigenvalues of random normal matrices. *Duke Math. J.* 159 31–81. MR2817648 https://doi.org/10.1215/00127094-1384782
- [2] AMEUR, Y., HEDENMALM, H. and MAKAROV, N. (2015). Random normal matrices and Ward identities. Ann. Probab. 43 1157–1201. MR3342661 https://doi.org/10.1214/13-AOP885
- [3] BACH, F., LACOSTE-JULIEN, S. and OBOZINSKI, G. (2012). On the equivalence between herding and conditional gradient algorithms. In *Proceedings of the International Conference on Machine Learning* (*ICML*).
- [4] BELLONI, A. and CHERNOZHUKOV, V. (2009). On the computational complexity of MCMC-based estimators in large samples. Ann. Statist. 37 2011–2055. MR2533478 https://doi.org/10.1214/08-AOS634
- [5] BERMAN, R. J. (2009). Bergman kernels for weighted polynomials and weighted equilibrium measures of  $\mathbb{C}^n$ . *Indiana Univ. Math. J.* **58** 1921–1946. MR2542983 https://doi.org/10.1512/iumj.2009.58.3644
- [6] BERMAN, R. J. (2009). Bergman kernels and equilibrium measures for line bundles over projective manifolds. Amer. J. Math. 131 1485–1524. MR2559862 https://doi.org/10.1353/ajm.0.0077
- [7] BERMAN, R. J. (2012). Sharp asymptotics for Toeplitz determinants and convergence towards the Gaussian free field on Riemann surfaces. *Int. Math. Res. Not. IMRN* 22 5031–5062. MR2997048 https://doi.org/10.1093/imrn/rnr229

- [8] BERMAN, R. J. (2014). Determinantal point processes and fermions on complex manifolds: Large deviations and bosonization. *Comm. Math. Phys.* 327 1–47. MR3177931 https://doi.org/10.1007/s00220-014-1891-6
- BERMAN, R. J. (2018). K\u00e4hler-Einstein metrics, canonical random point processes and birational geometry. In Algebraic Geometry: Salt Lake City 2015. Proc. Sympos. Pure Math. 97 29–73. Amer. Math. Soc., Providence, RI. MR3821145
- [10] BOGAERT, I. (2014). Iteration-free computation of Gauss–Legendre quadrature nodes and weights. SIAM J. Sci. Comput. 36 A1008–A1026. MR3209728 https://doi.org/10.1137/140954969
- [11] BRASS, H. and PETRAS, K. (2011). Quadrature Theory: The Theory of Numerical Integration on a Compact Interval. Mathematical Surveys and Monographs 178. Amer. Math. Soc., Providence, RI. MR2840013 https://doi.org/10.1090/surv/178
- [12] BREUER, J. and DUITS, M. (2016). Universality of mesoscopic fluctuations for orthogonal polynomial ensembles. *Comm. Math. Phys.* 342 491–531. MR3459158 https://doi.org/10.1007/s00220-015-2514-6
- [13] BREUER, J. and DUITS, M. (2017). Central limit theorems for biorthogonal ensembles and asymptotics of recurrence coefficients. J. Amer. Math. Soc. 30 27–66. MR3556288 https://doi.org/10.1090/jams/854
- [14] BRIOL, F.-X., OATES, C., GIROLAMI, M. and OSBORNE, M. A. (2015). Frank–Wolfe Bayesian quadrature: Probabilistic integration with theoretical guarantees. In Advances in Neural Information Processing Systems (NIPS) 1162–1170.
- [15] BRIOL, F.-X., OATES, C. J., GIROLAMI, M., OSBORNE, M. A. and SEJDINOVIC, D. (2019). Probabilistic integration: A role in statistical computation? *Statist. Sci.* 34 1–22. MR3938958 https://doi.org/10. 1214/18-STS660
- [16] CHEN, Y., WELLING, M. and SMOLA, A. (2010). Super-samples from kernel herding. In *Proceedings of the Conference on Uncertainty in Artificial Intelligence (UAI).*
- [17] DALEY, D. J. and VERE-JONES, D. (1988). An Introduction to the Theory of Point Processes. Springer Series in Statistics. Springer, New York. MR0950166
- [18] DAVIS, P. J. and RABINOWITZ, P. (1984). Methods of Numerical Integration, 2nd ed. Computer Science and Applied Mathematics. Academic Press, Orlando, FL. MR0760629
- [19] DELYON, B. and PORTIER, F. (2016). Integral approximation by kernel smoothing. *Bernoulli* 22 2177– 2208. MR3498027 https://doi.org/10.3150/15-BEJ725
- [20] DIACONIS, P. and EVANS, S. N. (2001). Linear functionals of eigenvalues of random matrices. *Trans. Amer. Math. Soc.* 353 2615–2633. MR1828463 https://doi.org/10.1090/S0002-9947-01-02800-8
- [21] DICK, J., KUO, F. Y. and SLOAN, I. H. (2013). High-dimensional integration: The quasi-Monte Carlo way. Acta Numer. 22 133–288. MR3038697 https://doi.org/10.1017/S0962492913000044
- [22] DICK, J. and PILLICHSHAMMER, F. (2010). Digital Nets and Sequences: Discrepancy Theory and Quasi-Monte Carlo Integration. Cambridge Univ. Press, Cambridge. MR2683394 https://doi.org/10.1017/ CBO9780511761188
- [23] DOUC, R., MOULINES, E. and STOFFER, D. S. (2014). Nonlinear Time Series: Theory, Methods, and Applications with R Examples. Chapman & Hall/CRC Texts in Statistical Science Series. CRC Press/CRC, Boca Raton, FL. MR3289095
- [24] FINK, M., NIEDERER, S. A., CHERRY, E. M., FENTON, F. H., KOIVUMÄKI, J. T., SEEMANN, G., THUL, R., ZHANG, H., SACHSE, F. B. et al. (2011). Cardiac cell modelling: Observations from the heart of the cardiac physiome project. *Prog. Biophys. Mol. Biol.* **104** 2–21.
- [25] GAUSS, C. F. (1815). Methodus Nova Integralium Valores per Approximationem Inveniendi. Heinrich Dietrich, Göttingen.
- [26] GAUTIER, G., BARDENET, R. and VALKO, M. (2018). DPPy: Sampling determinantal point processes with Python. Preprint. Available at arXiv:1809.07258.
- [27] GAUTSCHI, W. (2004). Orthogonal Polynomials: Computation and Approximation. Numerical Mathematics and Scientific Computation. Oxford Univ. Press, New York. MR2061539
- [28] GAUTSCHI, W. (2009/10). How sharp is Bernstein's inequality for Jacobi polynomials? *Electron. Trans. Numer. Anal.* 36 1–8. MR2779993
- [29] GAUTSCHI, W. and VARGA, R. S. (1983). Error bounds for Gaussian quadrature of analytic functions. SIAM J. Numer. Anal. 20 1170–1186. MR0723834 https://doi.org/10.1137/0720087
- [30] GLASER, A., LIU, X. and ROKHLIN, V. (2007). A fast algorithm for the calculation of the roots of special functions. SIAM J. Sci. Comput. 29 1420–1438. MR2341794 https://doi.org/10.1137/06067016X
- [31] GLYNN, P. W. and SZECHTMAN, R. (2002). Some new perspectives on the method of control variates. In *Monte Carlo and Quasi-Monte Carlo Methods*, 2000 (*Hong Kong*) 27–49. Springer, Berlin. MR1958845
- [32] GOLUB, G. H. and VAN LOAN, C. F. (2013). Matrix Computations, 4th ed. Johns Hopkins Studies in the Mathematical Sciences. Johns Hopkins Univ. Press, Baltimore, MD. MR3024913

- [33] HALE, N. and TOWNSEND, A. (2013). Fast and accurate computation of Gauss–Legendre and Gauss–Jacobi quadrature nodes and weights. SIAM J. Sci. Comput. 35 A652–A674. MR3033086 https://doi.org/10. 1137/120889873
- [34] HARDY, A. (2015). Average characteristic polynomials of determinantal point processes. Ann. Inst. Henri Poincaré Probab. Stat. 51 283–303. MR3300971 https://doi.org/10.1214/13-AIHP572
- [35] HOUGH, J. B., KRISHNAPUR, M., PERES, Y. and VIRÁG, B. (2006). Determinantal processes and independence. *Probab. Surv.* **3** 206–229. MR2216966 https://doi.org/10.1214/154957806000000078
- [36] HUSZÁR, F. and DUVENAUD, D. (2012). Optimally-weighted herding is Bayesian quadrature. In *Uncertainty in Artificial Intelligence (UAI)*.
- [37] JOHANSSON, K. (1997). On random matrices from the compact classical groups. Ann. of Math. (2) 145 519–545. MR1454702 https://doi.org/10.2307/2951843
- [38] JOHANSSON, K. (1998). On fluctuations of eigenvalues of random Hermitian matrices. Duke Math. J. 91 151–204. MR1487983 https://doi.org/10.1215/S0012-7094-98-09108-6
- [39] JOHANSSON, K. (2006). Random matrices and determinantal processes. In *Mathematical Statistical Physics* 1–55. Elsevier, Amsterdam. MR2581882 https://doi.org/10.1016/S0924-8099(06)80038-7
- [40] JOHANSSON, K. and LAMBERT, G. (2018). Gaussian and non-Gaussian fluctuations for mesoscopic linear statistics in determinantal processes. Ann. Probab. 46 1201–1278. MR3785588 https://doi.org/10. 1214/17-AOP1178
- [41] KALLENBERG, O. (1997). Foundations of Modern Probability. Probability and Its Applications (New York). Springer, New York. MR1464694
- [42] KILLIP, R. and NENCIU, I. (2004). Matrix models for circular ensembles. Int. Math. Res. Not. 50 2665– 2701. MR2127367 https://doi.org/10.1155/S1073792804141597
- [43] KÖNIG, W. (2005). Orthogonal polynomial ensembles in probability theory. Probab. Surv. 2 385–447. MR2203677 https://doi.org/10.1214/154957805100000177
- [44] KRIECHERBAUER, T. and SHCHERBINA, M. (2010). Fluctuations of eigenvalues of matrix models and their applications. Preprint. Available at arXiv:1003.6121.
- [45] KUIJLAARS, A. B. J., MCLAUGHLIN, K. T.-R., VAN ASSCHE, W. and VANLESSEN, M. (2004). The Riemann–Hilbert approach to strong asymptotics for orthogonal polynomials on [-1, 1]. Adv. Math. 188 337–398. MR2087231 https://doi.org/10.1016/j.aim.2003.08.015
- [46] KULESZA, A. and TASKAR, B. (2012). Determinantal Point Processes for Machine Learning. Foundations and Trends in Machine Learning.
- [47] LAMBERT, G. (2015). CLT for biorthogonal ensembles and related combinatorial identities. Preprint. Available at arXiv:1511.06121.
- [48] LAMBERT, G. (2018). Mesoscopic fluctuations for unitary invariant ensembles. *Electron. J. Probab.* 23 Paper No. 7. MR3771744 https://doi.org/10.1214/17-EJP120
- [49] LAVANCIER, F., MØLLER, J. and RUBAK, E. (2015). Determinantal point process models and statistical inference. J. R. Stat. Soc. Ser. B. Stat. Methodol. 77 853–877. MR3382600 https://doi.org/10.1111/ rssb.12096
- [50] LIU, Q. and LEE, J. D. (2017). Black-box importance sampling. Proc. Mach. Learn. Res. 54 952–961.
- [51] LYONS, R. (2003). Determinantal probability measures. Publ. Math. Inst. Hautes Études Sci. 98 167–212. MR2031202 https://doi.org/10.1007/s10240-003-0016-0
- [52] LYONS, R. and PERES, Y. (2016). Probability on Trees and Networks. Cambridge Series in Statistical and Probabilistic Mathematics 42. Cambridge Univ. Press, New York. MR3616205 https://doi.org/10. 1017/9781316672815
- [53] MACCHI, O. (1975). The coincidence approach to stochastic point processes. Adv. in Appl. Probab. 7 83– 122. MR0380979 https://doi.org/10.2307/1425855
- [54] MØLLER, J., NIELSEN, M., PORCU, E. and RUBAK, E. (2015). Determinantal point process models on the sphere. Research Report 13, Centre for Stochastic Geometry and Advanced Bioimaging.
- [55] O'HAGAN, A. (1991). Bayes–Hermite quadrature. J. Statist. Plann. Inference 29 245–260. MR1144171 https://doi.org/10.1016/0378-3758(91)90002-V
- [56] OATES, C. and GIROLAMI, M. (2016). Control functionals for quasi-Monte Carlo integration. In Proceeddings of the International Conference on Artificial Intelligence and Statistics.
- [57] OATES, C. J., GIROLAMI, M. and CHOPIN, N. (2017). Control functionals for Monte Carlo integration. J. R. Stat. Soc. Ser. B. Stat. Methodol. 79 695–718. MR3641403 https://doi.org/10.1111/rssb.12185
- [58] OLVER, S., NADAKUDITI, R. R. and TROGDON, T. (2015). Sampling unitary ensembles. *Random Matrices Theory Appl.* 4 1550002. MR3334666 https://doi.org/10.1142/S2010326315500021
- [59] OWEN, A. B. (1997). Scrambled net variance for integrals of smooth functions. Ann. Statist. 25 1541–1562. MR1463564 https://doi.org/10.1214/aos/1031594731
- [60] OWEN, A. B. (2008). Local antithetic sampling with scrambled nets. Ann. Statist. 36 2319–2343. MR2458189 https://doi.org/10.1214/07-AOS548

- [61] PASTUR, L. (2006). Limiting laws of linear eigenvalue statistics for Hermitian matrix models. J. Math. Phys. 47 103303. MR2268864 https://doi.org/10.1063/1.2356796
- [62] PEET, M. M. (2009). Exponentially stable nonlinear systems have polynomial Lyapunov functions on bounded regions. *IEEE Trans. Automat. Control* 54 979–987. MR2518110 https://doi.org/10.1109/ TAC.2009.2017116
- [63] POPESCU, I. (2009). General tridiagonal random matrix models, limiting distributions and fluctuations. Probab. Theory Related Fields 144 179–220. MR2480789 https://doi.org/10.1007/s00440-008-0145-y
- [64] PURVES, D., SCHARLEMANN, J. P. W., HARFOOT, M., NEWBOLD, T., TITTENSOR, D. P., HUTTON, J. and EMMOTT, S. (2013). Ecosystems: Time to model all life on Earth. *Nature* 493 295–297.
- [65] RASMUSSEN, C. E. and WILLIAMS, C. K. I. (2006). Gaussian Processes for Machine Learning. Adaptive Computation and Machine Learning. MIT Press, Cambridge, MA. MR2514435
- [66] RIDER, B. and VIRÁG, B. (2007). The noise in the circular law and the Gaussian free field. Int. Math. Res. Not. IMRN 2 Art. ID rnm006. MR2361453 https://doi.org/10.1093/imrn/rnm006
- [67] ROBERT, C. P. and CASELLA, G. (2004). Monte Carlo Statistical Methods, 2nd ed. Springer Texts in Statistics. Springer, New York. MR2080278 https://doi.org/10.1007/978-1-4757-4145-2
- [68] SAFF, E. B. and TOTIK, V. (1997). Logarithmic Potentials with External Fields. Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences] 316. Springer, Berlin. MR1485778 https://doi.org/10.1007/978-3-662-03329-6
- [69] SCARDICCHIO, A., ZACHARY, C. E. and TORQUATO, S. (2009). Statistical properties of determinantal point processes in high-dimensional Euclidean spaces. *Phys. Rev. E* (3) **79** 041108. MR2551206 https://doi.org/10.1103/PhysRevE.79.041108
- [70] SCHERVISH, M. J. (1995). Theory of Statistics. Springer Series in Statistics. Springer, New York. MR1354146 https://doi.org/10.1007/978-1-4612-4250-5
- [71] SHCHERBINA, M. (2013). Fluctuations of linear eigenvalue statistics of β matrix models in the multi-cut regime. J. Stat. Phys. 151 1004–1034. MR3063494 https://doi.org/10.1007/s10955-013-0740-x
- [72] SIMON, B. (2011). Szegő's Theorem and Its Descendants: Spectral Theory for L<sup>2</sup> Perturbations of Orthogonal Polynomials. M. B. Porter Lectures. Princeton Univ. Press, Princeton, NJ. MR2743058
- [73] SOSHNIKOV, A. (2000). Determinantal random point fields. Russian Math. Surveys 55 923–975.
- SOSHNIKOV, A. (2002). Gaussian limit for determinantal random point fields. Ann. Probab. 30 171–187. MR1894104 https://doi.org/10.1214/aop/1020107764
- [75] STAHL, H. and TOTIK, V. (1992). General Orthogonal Polynomials. Encyclopedia of Mathematics and Its Applications 43. Cambridge Univ. Press, Cambridge. MR1163828 https://doi.org/10.1017/ CBO9780511759420
- [76] SZEGŐ, G. (1975). Orthogonal Polynomials, 4th ed. Colloquium Publications, XXIII. Amer. Math. Soc., Providence, RI. MR0372517
- [77] TORQUATO, S., SCARDICCHIO, A. and ZACHARY, C. E. (2008). Point processes in arbitrary dimension from fermionic gases, random matrix theory, and number theory. J. Stat. Mech. Theory Exp. 11 P11019. MR2461401 https://doi.org/10.1088/1742-5468/2008/11/p11019
- [78] TROTTA, R. (2008). Bayes in the sky: Bayesian inference and model selection in cosmology. Contemp. Phys. 49 71–104. https://doi.org/10.1080/00107510802066753
- [79] XIANG, S. (2016). An improved error bound on Gauss quadrature. Appl. Math. Lett. 58 42–48. MR3482993 https://doi.org/10.1016/j.aml.2016.01.015
- [80] XIANG, S. and BORNEMANN, F. (2012). On the convergence rates of Gauss and Clenshaw–Curtis quadrature for functions of limited regularity. SIAM J. Numer. Anal. 50 2581–2587. MR3022232 https://doi.org/10.1137/120869845